



FINANCIAL STATEMENTS | ANNUAL REPORT

2019-20

Contents

Directors' Report	•
Governance Statement	5
Statement of Profit or Loss	
and Other Comprehensive Income	8
Statement of Financial Position	S
Statement of Cash Flows	10
Statement of Changes in Accumulated Funds	1
Notes to the Financial Statements	12
Board's Declaration	24
Independent Auditor's Report to the	
Members of the Western Australian	
Cricket Association Limited	25
Independence Declaration	28
Constitution	29

This Annual Report is prepared for the 23 September 2020 Annual General Meeting of Members. Copies of the 2019-20 edition of The Western Cricketer will be available on the night of the Annual General Meeting and thereafter on application from the WACA Membership Office.

Directors' Report

For the year ended 30 June 2020

The Directors present their report, together with the financial statements, on the Western Australian Cricket Association Limited ("company") for the financial year ended 30 June 2020.

Directors

Name	Category
The Hon TK (Terrence) Waldron (Chair)	Member Elected Director (Chair from the 25 September 2019)
The Hon Dr KC (Ken) Michael AC (Chair)	Member Elected Director (to 25 September 2019)
Dr VA (Vanessa) Guthrie (Deputy Chair)	Board Appointed Director (Re-appointed 25 September 2019)
TF (Thomas) Percy QC	Member Elected Director
MR (Michael) Veletta	Member Elected Director (Re-elected 25 September 2019)
DJ (Darren) Wates	Member Elected Director
GM (Graeme) Wood	Member Elected Director (Elected 25 September 2019)
ED (Eva) Skira AM	Board Appointed Director
DJ (David) Bailey	Board Appointed Director
AJ (Avril) Fahey	Cricket Appointed Director
CK (Christian) Bauer	Cricket Appointed Director (Re-appointed 25 September 2019)

The above-named Directors held office during the whole financial year and since the end of the financial year unless otherwise stated.

The Board thanks Hon Dr Ken Michael AC for his enormous contribution over many years to both the WACA and to Cricket.

No Director has received any remuneration from the Association or the company during the year or after the year-end.

The company constitution provides for up to ten Directors (five elected by Members, two by cricket affiliates and up to three appointed by the Board).

Company Secretary

RI (Richard) Marshall

Nature of Operations and Principal Activities

Western Australian Cricket Association Limited is a not-for-profit company and any surplus income or property must be applied to promote its principal activities.

The company's principal activities are the promotion, development and governance of cricket at all levels, from grassroots to elite, in Western Australia.

The company is a member of the national body, Cricket Australia ("CA") and works closely with CA on the development of strategies to improve the delivery of cricket programs.

Strategic Plan

The company operates in line with its a strategic plan (2017-2022), developed in conjunction with CA.

Purpose: To inspire all Western Australians to love cricket.

Vision: To be leaders in Australian Sport.

The strategic plan has three pillars:

- Fans; to deliver what our Members and fans want through a focus on innovation, customer service and engaging content;
- Participants and Volunteers; to create the best participation experiences to grow junior cricket and ensure best practice
 model for senior community competitions;
- · Elite Teams; to deliver the best high-performance system in Western Australian sport.

The objectives which span across all three pillars:

- We will be the leading sport for women and girls and all diversity groups;
- · We will promote cricket to inspire love of the game in WA;
- · We will use technology to deliver great experiences for Members, fans, participants and volunteers;
- We will maximise long-term sustainable revenue to drive investment in WA cricket.

The strategy is supported by:

- Our People the platform of the organisation;
- Our Stakeholders building strong, trusting relationships;
- Our Governance delivering best practice at all levels;
- Our Ground a community destination.

The strategic plan is in the process of being refreshed and updated to take into account the current environment and all the challenges that lay ahead.

Review of Operations

The WACA was in a fortunate position to complete all cricket games bar the last Sheffield Shield game of the season resulting in revenue not being materially impacted by the COVID-19 global health pandemic which commenced in March 2020. To combat COVID-19 and protect cash reserves into the future the WACA developed a business continuity plan to protect cash reserves, the assets of the WACA and all forms of cricket (grassroots, community cricket, diversity and inclusion programs and elite performance across the State). With strong cash reserves held by the WACA, a strategic, timed and people first approach was taken to reset the business and set up for the challenges that lay ahead with the potential impact of reduced cricket content, the inability to have mass gatherings, the various quarantine laws to enter each state on top of poor global economic conditions.

To protect cash reserves in the last quarter of the financial year the WACA reduced expenditure and ceased cricket programs, while ensuring all employees had security of employment during this period to enable the WACA to grow reserves and battle the pandemic in 2020-21 and beyond.

After a thorough review of the business to combat COVID-19, the WACA made the difficult decision to reduce its cost base by 31% in June 2020. This was needed to protect the WACA and combat the challenges that lay ahead with the unknown impact to revenue streams in the 2020-21 financial year and beyond.

The potential impact to revenue due to COVID-19 in 2020-21 are:

- Reduced Cricket Australia funding by up to 25%;
- The potential changing cricket fixtures (for both International matches and the Big Bash League) due to COVID-19 Government restrictions:
- Estimated lower (with the potential of no) crowd attendance;
- · Potential reduction in membership and sponsorship revenue if cricket content is compromised;

Even with the impact of COVID-19 in the 2019-20 financial year, the WACA produced a revenue figure of \$38,567,364. This revenue was up \$990,183 (or 2.6%) on last year and was derived from several different sources, including:

- The WACA Members area at Optus Stadium and the WACA Ground resulted in membership revenue of \$6,118,128 in 2020 (compared to \$6,995,079 in 2019);
- The Big Bash League which generated ticketing, marketing, membership and catering revenue;
- The success of the WACA Ground function and events business growing sales from \$575,936 in FY19 to \$1,316,582 in FY20;

The revenue noted above supported investment in several areas, most notably cricket operations, which incorporates the promotion of cricket at all levels.

The move of cricket to Optus Stadium is an important long-term opportunity to increase the profile of cricket and enable more fans to watch cricket at a bigger modern venue. The seven home Big Bash League games had an average attendance of 26,935 (compared to 30,133 in 2019 and 21,493 in 2018 - the last year at the WACA Ground).

With the move of major matches to Optus Stadium more cricket fans have the opportunity to become a WACA Member. Several different tiers of membership, with different seating options and premium products are available. The membership and guest card numbers this year were 14,597 down on 15,975 last year. These Members and guests enjoyed the WACA Members Reserve area at Optus Stadium and the WACA Ground with excellent viewing with Member only bars and facilities.

The second Test Match at Optus Stadium was played against New Zealand in extreme heat in December 2019 in front of 64,978 fans. This was a record crowd in Western Australia against New Zealand with only 46,800 attending the Test Match played at the WACA Ground in 1985.

The WACA has its own catering, function and events business to use the WACA Ground to generate revenue for our broader business. This year the WACA Ground generated record sales of \$1,316,582 (up \$740,646 or 129% on last year). This was an outstanding result considering the impact of COVID-19 and the WACA Ground being closed for the last quarter of the financial year and generating no sales in that period.

With the continued investment of cricket in WA, the result was an operating surplus of \$660,590 before depreciation and interest expense and \$250,588 operating surplus after interest and depreciation. This surplus has helped protect cash reserves with a closing cash balance of \$15,465,113 and helps the WACA to combat all the COVID-19 challenges that lay ahead.

Likely Developments

In December 2019, the Federal Government confirmed a \$30,000,000 funding commitment towards the WACA Ground Improvement Project. Following this announcement, the WACA was asked to submit a Community Development Grant application to finalise the funding arrangement for this commitment. The WACA is currently awaiting the Funding Agreement from the Federal Government.

In August 2020, the State Government announced it would match the \$30,000,000 Federal Government funding commitment towards the Project. The funding forms part of a \$130,000,000 package to upgrade sports facilities as part of the WA Recovery Plan.

The WACA is currently exploring the feasibility of an opportunity to partner with the City of Perth and include an aquatic facility at the WACA Ground. The WACA is in the process of working with the City of Perth to determine if a 50 meter pool, a learn to swim pool and a leisure slide can be integrated into the design of the WACA Ground Improvement Project.

A number of complex matters including management, an appropriate operating model and tenure need to be determined before an informed decision can be made by the WACA and the City of Perth. Further, the State Government funding commitment being contingent on the City of Perth providing a funding commitment towards the proposed aquatic facilities will be discussed further once the WACA receives details of the Funding Agreement from the State Government.

Cricket Australia has recently confirmed that their \$4,000,000 funding commitment towards the Project remains secure and the WACA will raise \$10,000,000 towards the Project through the WA Cricket Foundation.

The next 3 months involves finalising detailed design and planning for the WACA Ground Improvement Project, determining the feasibility of an aquatic facility as part of the Project and ensuring that the Project is 'shovel ready'. The WACA has employed a Project Client Lead to ensure this work being undertaken is expedited effectively and efficiently.

The WACA will continue to inform WACA Members on the progress of this exciting Project.

Significant matters since the year-end

Other than the changing impact and the uncertainty of the COVID-19 pandemic, there has not arisen any item, transaction or event of a material nature, likely in the opinion of the Directors, to materially affect the operations or the affairs of the company in future years.

Board Meeting Attendance

A total of thirteen (13) Board meetings were held during the financial year.

Name	Meetings Attended	
The Hon TK (Terrence) Waldron (Chair)	11	
The Hon Dr KC (Ken) Michael AC (Chair)	3/3	
Dr VA (Vanessa) Guthrie (Deputy Chair)	13	
TF (Thomas) Percy QC	12	
MR (Michael) Veletta	11	
DJ (Darren) Wates	13	
GM (Graeme) Wood	10/10	
ED (Eva) Skira AM	11	
DJ (David) Bailey	12	
AJ (Avril) Fahey	11	
CK (Christian) Bauer	13	

Board Committees

The following Directors served on the various Board Committees during the financial year.

Governance Committee: Dr Vanessa Guthrie (Chair), Tom Percy QC, Graeme Wood and Christian Bauer.

Nominations and Remuneration Committee: Hon Terry Waldron (Chair), David Bailey, Dr Vanessa Guthrie and Avril Fahey.

Audit and Risk Committee: David Bailey (Chair), Avril Fahey, Michael Veletta and Eva Skira.

WACA Ground Improvement Project Control Group: Darren Wates (Chair), Hon Terry Waldron and Christian Bauer.

WA Aboriginal Cricket Advisory Committee: Hon Terry Waldron (Co-Chair).

Governance

Please refer to the separate Governance Statement set out on the following page.

Dated at Perth this 12th day of August 2020.

Signed in accordance with a resolution of the Board:

The Hon TK Waldron | Chair

DJ Bailey | Chair, Audit & Risk Committee

Governance Statement

This Governance Statement reflects the new Constitution which came into effect on 5 July 2018.

Western Australian Cricket Association Limited ("WACA") is a public company limited by guarantee, incorporated under the *Corporations Act 2001*. Ultimate responsibility for the governance of the company rests with the Board of Directors. This Governance Statement outlines how the Board meets that responsibility.

The Board's Role

The Board's primary role is to ensure that WACA's strategic direction is in accordance with the objectives stated in the constitution and that the activities are in line with achieving the strategic plan. The current strategic plan has three pillars; Fans, Participants and Volunteers, Elite Teams, designed to deliver the following objectives:

- To be the leading sport for women and girls and all diversity groups;
- To promote cricket to inspire love of the game in WA;
- To use technology to deliver great experiences for Members, fans, participants and volunteers;
- To maximise long-term sustainable revenue to drive investment in WA cricket.

Supporting strategies behind all three pillars cover; Our People, Our Stakeholders, Our Governance and Our Ground.

The Board fulfils its primary role by:

- Setting the strategic direction for the WACA, assessing progress towards strategic objectives and approving policies to work towards those objectives;
- Selecting, appointing, guiding and monitoring the performance of the CEO;
- · Overseeing the development and maintenance of the WACA's Behaviour Statements;
- Approving and monitoring the annual budget and ongoing forecasts;
- Monitoring the WACA's financial performance against budgets and forecasts;
- Overseeing and approving strategies and policies to ensure appropriate risk management;
- Ensuring appropriate delegations to management are in place;
- Appointing Board Committees as appropriate from time to time;
- Acting as an advocate for the WACA as appropriate.

The Board responsibilities are set out in the WACA Governance charter.

Management's Responsibility

The Board has formally delegated responsibility for WACA's day-to-day operations to the CEO and General Managers (Leadership Team).

The WACA's management team comprises the CEO and the General Management team covering; High Performance, Community Cricket, Membership, Business Services, Commercial, Marketing and Communication, WACA Ground and Support Venues, People and Culture and Public Policy/ Government Relations. The CEO provides the leadership of the management team and the organisation. The CEO is responsible for achieving the targets set out in the strategic plan and is authorised by the Board to put in place policies and practices, take decisions and actions and initiate activities to achieve those results.

Board Meetings and Board Committees

The Board meets as often as required to professionally discharge its duties. This is generally monthly with additional meetings scheduled as required. Workshops or forums are held periodically to provide opportunities for further Board engagement in strategic and other important issues. The CEO and the General Manager – Business Services/Company Secretary participate in all Board meetings, as directed by the Board Chair. General Managers present to the Board from time to time as determined by the Board.

Agenda setting is a collaborative process with input from the Board Chair, Committee Chairs, CEO and Company Secretary with final approval by the Board Chair.

Board Committees are established from time to time as required.

Presently the following Board Committees are in place and active:

- Governance Committee;
- Nominations & Remuneration Committee;
- · Audit and Risk Committee;
- WACA Ground Improvement Project Control Group;
- WA Aboriginal Cricket Advisory Committee;

Each Board Committee has a Director as Chair and includes other Directors. The Board does not delegate major decisions to its Committees but charges them with the responsibility to consider the detailed issues and to make appropriate recommendations to the Board.

Each Board Committee operates under a charter approved by the Board. These charters are reviewed annually and updated as necessary.

Governance Committee

The Governance Committee considers and reviews the governance of the organisation on behalf of the Board and makes recommendations to the Board. The constitution, governance charter, policy framework and Board policies are reviewed by the Governance Committee bi-annually. The Committee consists of Board Directors.

Nominations and Remuneration Committee

The Nominations and Remuneration Committee ("NRC") develops and monitors the Board skills matrix, which is used when considering the skills required from people nominating to join the Board. The NRC manages the nomination process on behalf of and makes recommendations to the Board. The NRC also develops and monitors the remuneration policy and succession policies regarding the CEO and General Managers. The Committee consists of Board Directors, including the WACA Chair.

Audit and Risk Committee

The Audit and Risk Committee provides oversight and monitors the financial systems, financial reporting, risk management and audit functions. Recommendations are made to the Board in relation to approval of finance related documents such as the annual budget, ongoing forecasts and annual financial report as well as the risk management framework and statements. The Committee consists of Board Directors and independent experts, appointed from time to time.

WACA Ground Improvement Project Control Group

The WACA Ground ("Ground") is a vital, valuable and strategic asset owned by the WACA.

The WACA Ground Improvement Project Control Group ("GIPCG") considers and reviews the improvement of the Ground in the context of the WACA's strategic plans. The GIPCG oversees and monitors the development of funding strategies and applications to government and other organisations to create a sustainable community and sporting hub with cricket at its heart. The GIPCG reports to the Board at each Board meeting and as appropriate. The composition of the GIPCG includes Directors (including the Chair of the GIPCG), CEO, management, consultants with expertise in land development and representatives from stakeholders with expertise in facilities.

Other Committees

The WA Aboriginal Cricket Advisory Committee and WA Education Advisory Committee were formed during 2017 to provide a more effective method of governance for these sections of the cricket community. Previously both sections were governed by their respective affiliated bodies.

The Board may establish other Committees and working groups for considering and advising on specific matters. These Committees report to the Board during the course of normal Board meetings.

Board Directors

All Board Directors are non-executive Directors and receive no remuneration for their services.

WACA's constitution provides an indemnity to Directors. Management Liability insurance is in place. The company's constitution specifies that:

- There must be no fewer than seven and no more than ten Directors (five Directors elected by WACA Members, two
 Directors elected by cricket affiliates and up to three Directors appointed by the Board to complement skills and to fill
 any skill gaps);
- · No employees of the company, including the CEO, can be a Director of the company;
- Directors are appointed for a maximum of three terms of three years.

Board and Committee members receive written advice of the terms and conditions of their appointment and complete a structured induction programme when first appointed. The performance of individual Board and Committee members and the Board and Board Committees is assessed annually.

The Chair

The Chair of the Board is elected by the Board of Directors. The key internal roles of the Chair are to:

- Ensure the Board provides vision and guidance to the WACA;
- · Ensure Board meetings are effective;
- Ensure the Board considers matters in a timely, transparent manner;
- Guide the effectiveness and development of the Board and individual Directors.

Externally, the Chair acts as spokesperson for WACA in conjunction with the CEO and consults and communicates with stakeholders.

Risk Management

The Board has overall oversight of risk management and approves a risk appetite statement to set the overall approach to various risks facing the WACA. A risk management policy is a part of the WACA Governance Charter.

Management is responsible for the identification of risks, the evaluation of such risks and the development of plans to manage the risks. Risk management is monitored by the Audit and Risk Committee. These risks are rated and assigned to General Managers as the risk owners who are responsible for the day to day management of each risk.

Code of Conduct

Board Directors, the CEO, General Managers and staff are expected to comply with the WACA's Code of Conduct, which have been developed in conjunction with Cricket Australia and other State Cricket Associations.

Board Directors, management and staff are expected to comply with all relevant laws, the codes of conduct of relevant professional bodies, and act with integrity, compassion, fairness and honesty.

Board Directors, Board Committee members and staff are made aware of WACA's ethical standards, code of conduct and conflicts of interest policy during their induction to the organisation and are provided with a copy of these documents at that time.

Any potential conflicts at the Board or Committee level are disclosed and noted on a register of interests, and this is disclosed at every meeting of the Board, with updates being made as and when new or changed interests are advised.

Board Reviews

The Board regularly reviews its performance. The purpose of the review is to identify issues relating to the skills, behaviour, relationships or practices that may be inhibiting the Board from being fully effective. The Board believes that constructive feedback helps it operate effectively.

Transparency with Stakeholders

WACA has many stakeholders, including our Members, staff, players, affiliates, clubs, volunteers, fans in the broader community, Cricket Australia, the government agencies that provide funds and regulate our operations, and our suppliers. Transparent disclosure is a part of accountability to all our stakeholders.

The WACA prepares and make available a variety of reports, each aimed at providing the information necessary to improve accountability and transparency to all stakeholders. These include the Western Cricketer, the annual financial report and the strategic plan in addition to regular communications to Members and other stakeholders.

The Constitution and WACA Governance Charter are published on the WACA's website.

STATEMENT OF PROFIT OR LOSS AND OTHER COMPREHENSIVE INCOME

For the year ended 30 June 2020

	NOTES	2020 (\$)	2019 (\$)
REVENUE			
Revenue	2(i)	38,348,328	37,186,198
Other Income	2(ii)	219,036	390,983
Employee benefit expenses		(20,011,200)	(19,134,665)
Match & event expenses		(6,527,029)	(6,913,559)
Equipments & maintenance expense		(2,309,679)	(2,261,996)
Travel & entertainment expense		(2,038,458)	(2,257,066)
Contractors & consultants expense		(2,016,646)	(1,838,024)
Administration expense		(1,383,600)	(1,496,807)
Promotion & advertising expense		(988,585)	(966,755)
Other expenses		(1,323,278)	(1,525,009)
Grants to affiliate bodies		(1,308,300)	(1,221,877)
Interest expense		(12,403)	(700 440)
Depreciation		(397,599)	(308,440)
Profit for the period		250,588	(347,017)
Other Comprehensive Income / (Expenses)		-	-
Total Comprehensive Income / (Deficit)		250,588	(347,017)

The statement of profit or loss and other comprehensive income is to be read in conjunction with the notes to the financial statements set out on pages 12 to 23.

STATEMENT OF FINANCIAL POSITION

As at 30 June 2020

	NOTES	2020 (\$)	2019 (\$)
CURRENT ACCETS			
CURRENT ASSETS			
Cash and cash equivalents Trade and other receivables	5 6	15,465,113	15,811,715
Inventories	7	230,032 21,509	1,426,297 12,372
Other current assets	8	-	70,152
Total current assets		15,716,653	17,320,535
NON-CURRENT ASSETS			
Property, plant and equipment	9	1,807,326	1,135,233
Total non-current assets		1,807,326	1,135,233
Total assets		17,523,979	18,455,768
CURRENT LIABILITIES			
Trade and other payables	10	4,577,317	6,130,612
Lease liabilities	11	62,916	-
Provisions	12	1,279,828	1,231,283
Deferred income	13	225,831	282,703
Total current liabilities		6,145,892	7,644,599
NON-CURRENT LIABILITIES			
Lease liabilities	11	408,861	-
Provisions	12	200,973	293,505
Total non-current liabilities		609,834	293,505
Total liabilities		6,755,725	7,938,104
NET ASSETS		10,768,254	10,517,664
ACCUMULATED FUNDS			
Accumulated surplus		10,768,254	10,517,664
Total accumulated funds		10,768,254	10,517,664
Total accumulated funds		10,700,234	10,517,004

The statement of financial position is to be read in conjunction with the notes to the financial statements set out on pages 12 to 23.

STATEMENT OF CASH FLOWS

For the year ended 30 June 2020

		NOTES	2020 (\$)	2019 (\$)
Cash receipts in the course of operations Cash payments in the course of operations Interest received Net cash provided by operating activities CASH FLOWS FROM INVESTING ACTIVITIES Proceeds on disposal of property, plant and equipment Payments for property, plant and equipment Net cash used in investing activities CASH FLOWS FROM FINANCING ACTIVITIES Lease liability repayment Net cash (used in) financing activities Cash at the beginning of the financial year 39,487,721 (39,443,041) (34,449,383) (39,443,041) (34,449,383) (39,487,721 (39,443,041) (34,449,383) (346,611) (346,611) (346,614) (346,614) (346,601) (346,601) (346,601) (3476,532 (39,443,041) (344,49,383) (39,487,721 (39,443,041) (344,49,383) (39,487,721 (39,443,041) (344,49,383) (39,487,721 (39,443,041) (344,49,383) (39,487,721 (39,443,041) (344,49,383) (39,487,721 (39,443,041) (344,49,383) (39,487,721 (39,443,041) (344,49,383) (346,615) (346,614) (346,614) (346,601) (346,601) (346,601) (3476,532 (346,601) (3476,532				
Cash payments in the course of operations Interest received Net cash provided by operating activities CASH FLOWS FROM INVESTING ACTIVITIES Proceeds on disposal of property, plant and equipment Payments for property, plant and equipment Net cash used in investing activities CASH FLOWS FROM FINANCING ACTIVITIES Lease liability repayment Net cash (used in) financing activities (64,614) Net (decrease)/increase in cash held Cash at the beginning of the financial year (39,443,041) (24,449,383) 219,036 381,035 263,715 2,033,803 10,909 (545,702) (268,179) (257,270) (257,270) (64,614) - (64,614) - (74,614) - (75,532 (75,811,714) (75,532) (75,811,714) (75,812,811,714)	CASH FLOWS FROM OPERATING ACTIVITIES			
Interest received Net cash provided by operating activities CASH FLOWS FROM INVESTING ACTIVITIES Proceeds on disposal of property, plant and equipment Payments for property, plant and equipment Net cash used in investing activities CASH FLOWS FROM FINANCING ACTIVITIES Lease liability repayment Net cash (used in) financing activities (64,614) Net (decrease)/increase in cash held Cash at the beginning of the financial year 219,036 381,035 2,033,803 10,909 10				
Net cash provided by operating activities CASH FLOWS FROM INVESTING ACTIVITIES Proceeds on disposal of property, plant and equipment Payments for property, plant and equipment Net cash used in investing activities CASH FLOWS FROM FINANCING ACTIVITIES Lease liability repayment Net cash (used in) financing activities (64,614) Net (decrease)/increase in cash held Cash at the beginning of the financial year				
CASH FLOWS FROM INVESTING ACTIVITIES Proceeds on disposal of property, plant and equipment Payments for property, plant and equipment (545,702) (268,179) Net cash used in investing activities (545,702) (257,270) CASH FLOWS FROM FINANCING ACTIVITIES Lease liability repayment (64,614) Net cash (used in) financing activities (64,614) Net (decrease)/increase in cash held (346,601) 1,776,532 Cash at the beginning of the financial year		15(ii)		
Proceeds on disposal of property, plant and equipment Payments for property, plant and equipment Net cash used in investing activities CASH FLOWS FROM FINANCING ACTIVITIES Lease liability repayment Net cash (used in) financing activities (64,614) Net (decrease)/increase in cash held Cash at the beginning of the financial year 10,909 (545,702) (268,179) (545,702) (257,270) (257,270) (257,270) (257,270) (257,270) (257,270) (257,270) (257,270) (257,270) (257,270)		10(11)		
Payments for property, plant and equipment (545,702) (268,179) Net cash used in investing activities (545,702) (257,270) CASH FLOWS FROM FINANCING ACTIVITIES Lease liability repayment (64,614) - Net cash (used in) financing activities (64,614) - Net (decrease)/increase in cash held (346,601) 1,776,532 Cash at the beginning of the financial year 15,811,714 14,035,182				
Net cash used in investing activities (545,702) (257,270) CASH FLOWS FROM FINANCING ACTIVITIES Lease liability repayment (64,614) - Net cash (used in) financing activities (64,614) - Net (decrease)/increase in cash held (346,601) 1,776,532 Cash at the beginning of the financial year 15,811,714 14,035,182			- (E4E 702)	
CASH FLOWS FROM FINANCING ACTIVITIES Lease liability repayment Net cash (used in) financing activities Net (decrease)/increase in cash held Cash at the beginning of the financial year (64,614) - (346,601) 1,776,532 15,811,714 14,035,182		-		
Lease liability repayment(64,614)-Net cash (used in) financing activities(64,614)-Net (decrease)/increase in cash held(346,601)1,776,532Cash at the beginning of the financial year15,811,71414,035,182	Net cash used in investing activities	-	(545,702)	(257,270)
Net cash (used in) financing activities(64,614)-Net (decrease)/increase in cash held(346,601)1,776,532Cash at the beginning of the financial year15,811,71414,035,182	CASH FLOWS FROM FINANCING ACTIVITIES			
Net (decrease)/increase in cash held (346,601) 1,776,532 Cash at the beginning of the financial year 15,811,714 14,035,182	Lease liability repayment		(64,614)	-
Cash at the beginning of the financial year 15,811,714 14,035,182	Net cash (used in) financing activities		(64,614)	-
Cash at the beginning of the financial year 15,811,714 14,035,182	Net (decrease)/increase in cash held		(346,601)	1,776,532
Cash at the end of the financial year 15(i) 15,465,113 15,811,714	Cash at the beginning of the financial year		15,811,714	14,035,182
	Cash at the end of the financial year	15(i)	15,465,113	15,811,714

The statement of cash flows is to be read in conjunction with the notes to the financial statements set out on pages 12 to 23.

STATEMENT OF CHANGES IN ACCUMULATED FUNDS

For the year ended 30 June 2020

	Accumulated Funds (\$
Balance at 30 June 2018	10,864,682
Fotal Comprehensive Deficit	(347,016
Balance at 30 June 2019	10,517,666
Fotal Comprehensive Deficit	250,588
Balance at 30 June 2020	10,768,254

The statement of changes in accumulated funds is to be read in conjunction with the notes to the financial statements set out on pages 12 to 23.

NOTES TO THE FINANCIAL STATEMENTS

For the year ended 30 June 2020

1. Statement of significant accounting policies

The significant policies which have been adopted in the preparation of this financial report are:

(a) Basis of preparation

The financial report is a general purpose financial report, which has been prepared in accordance with Australian Accounting Standards and the Corporations Act 2001 in a form appropriate for a sporting association.

It has been prepared on the basis of historical cost except where stated and explained in the accounting policies below.

The accounting policies have been consistently applied and, except where there is a change in accounting policy, are consistent with those of the previous year.

New and revised AASBs affecting amounts reported and/or disclosures in the financial statements

The Company has adopted all of the new and revised Standards and Interpretations issued by the Australian Accounting Standards Board (the AASB) that are relevant to its operations and effective for the current reporting period.

The Company has adopted all of the new and revised Standards and Interpretations issued by the Australian Accounting Standards Board that are relevant to their operations and are effective for the current financial reporting period, being the year ended 30 June 2020. New and revised standards and amendments thereof and interpretations effective for the current reporting period that are relevant to the company include:

- AASB 16 Leases
- AASB 15 Revenue from Contracts with Customers
- AASB 1058 Income of Not-for-Profit Entities

The adoption of new and revised standards and amendments has not affected the amounts reported for the prior years. A discussion on the impact of the adoption of AASB 16, AASB 1058 and AASB 15 is included below. The Company has not early adopted any other standard, interpretation or amendment that has been issued but is not yet effective.

AASB 1058 Income of Not-for-Profit Entities (AASB 1058) and AASB 15 Revenue from Contracts with Customers (AASB 15)

AASB 1058 clarifies and simplifies the income recognition requirements that apply to not-for-profit (NFP) entities, in conjunction with AASB 15. The new income recognition requirements shift the focus from a reciprocal/non-reciprocal basis to a basis of assessment that considers the enforceability of a contract and the specificity of performance obligations.

The core principle of the new income recognition requirements in AASB 1058 is when a NFP entity enters into transactions where the consideration to acquire an asset is significantly less than the fair value of the asset principally to enable the entity to further its objectives, the excess of the asset recognised (at fair value) over any 'related amounts' is recognised as income immediately.

An example of a 'related amount' is AASB 15 and in cases where there is an 'enforceable' contract with a customer with 'sufficiently specific' performance obligations, income is recognised when (or as) the performance obligations are satisfied under AASB 15, as opposed to immediate income recognition under AASB 1058

Under AASB 15, the Company recognises revenue when (or as) a performance obligation is satisfied, i.e. when 'control' of the goods or services underlying the particular performance obligation is transferred to the customer. The adoption of AASB 15/AASB 1058 did not have a material effect on the Company's financial statements as at 1 June 2020.

AASB 16 Leases

In the current year, the Company has applied AASB 16 Leases, which is effective for annual periods that begin on or after 1 January 2019.

AASB 16 introduces new or amended requirements with respect to lease accounting. It introduces significant changes to lessee accounting by removing the distinction between operating and finance lease and requiring the recognition of a right-of-use asset and a lease liability at commencement for all leases, except for short-term leases and leases of low value assets. Details of these new requirements are described in note 1. The impact of the adoption of AASB 16 on the Company's consolidated financial statements is described below.

The date of initial application of AASB 16 for the Company is 1 July 2019.

The Company has applied AASB 16 using the cumulative catch-up approach which:

- Requires the Company to recognise the cumulative effect of initially applying AASB 16 as an adjustment to the opening balance of retained earnings at the date of initial application.
- Does not permit restatement of comparatives.

Impact of the new definition of lease

The Company has made use of the practical expedient available on transition to AASB 16 not to reassess whether a contract is or contains a lease. Accordingly, the definition of a lease in accordance with AASB 117 Leases and Interpretation 4 Determining whether an Arrangement contains a Lease will continue to be applied to those contracts entered or modified before 1 July 2019.

The change in definition of a lease mainly relates to the concept of control. AASB 16 determines whether a contract contains a lease on the basis of whether the customer has the right to control the use of an identified asset for a period of time in exchange for consideration. This is in contrast to the focus on 'risks and rewards' in AASB 117 and Interpretation 4.

The Company applies the definition of a lease and related guidance set out in AASB 16 to all contracts entered into or changed on or after 1 July 2019. In preparation for the first-time application of AASB 16, the Company has carried out an implementation project. The project has shown that the new definition in AASB 16 will not significantly change the scope of contracts that meet the definition of a lease for the Company.

(a) Impact of the new definion of lease

(i) Former operation leases

AASB 16 changes how the Company accounts for leases previously classified as operating leases under AASB 117, which were off balance sheet.

Applying AASB 16, for all leases (except as noted below), the Company:

- Recognises right-of-use assets and lease liabilities in the consolidated statement of financial position, initially measured at the present value of the future lease payments
- · Recognises depreciation of right-of-use assets and interest on lease liabilities in profit or loss
- Separates the total amount of cash paid into a principal portion (presented within financing activities) and interest (presented within financing activities) in the consolidated statement of cash flows.

Lease incentives (e.g. rent-free period) are recognised as part of the measurement of the right-of-use assets and lease liabilities whereas under AASB 117 they resulted in the recognition of a lease incentive, amortised as a reduction of rental expenses generally on a straight-line basis. Under AASB 16, right-of-use assets are tested for impairment in accordance with AASB 136 Impairment of Assets.

For short-term leases (lease term of 12 months or less) and leases of low-value assets, the Company has opted to recognise a lease expense on a straight-line basis as permitted by AASB 16. This expense is presented within 'other expenses' in profit or loss.

Former finance leases

The main differences between AASB 16 and AASB 117 with respect to contracts formerly classified as finance leases is the measurement of the residual value guarantees provided by the lessee to the lessor. AASB 16 requires that the Company recognises as part of its lease liability only the amount expected to be payable under a residual value guarantee, rather than the maximum amount guaranteed as required by AASB 117. This change did not have effect on the Company's financial statements, as currently the Company does not have any finance lease.

Impact of Adoption of Lease

The weighted average lessee's incremental borrowing rate applied to lease liabilities at transition, i.e. 1 June 2019 was 2.70%.

(b) Financial impact of initial application of AASB 16

The following table shows the operating lease commitments disclosed applying AASB 16 at 1 July 2019.

Impact on assets, liabilities and equity as at 1 July 2019

	AASB 16 adjustments (\$)
Right-of-use assets	523,989
Net impact on total assets	523,989
Lease liabilities	523,989
Net impact on total liabilities	523,989

Standards and Interpretations in issue not yet adopted

At the date of authorisation of the financial statements, the Standards and Interpretations listed below were in issue but not yet effective.

Standard/Interpretation	Effective for annual reporting periods beginning on or after	Expected to be initially applied in the financial year ending
AASB 2018-7 Amendment to Australian Accounting Standards – Definition of Material	1 January 2020	30 June 2021
AASB 2019-1 Amendments to Australian Accounting Standards - References to the Conceptual Framework	1 January 2020	30 June 2021
AASB 2019-5 Amendments to Australian Accounting Standards - Disclosure of the Effect of New IFRS Standards Not Yet Issued in Australia	1 January 2020	30 June 2021

Management are yet to undertake a formal assessment of the impact of the accounting standards that are issued but not yet effective, but the impact on the Company is anticipated to be immaterial as the majority do not impact its current operations.

(b) Statement of compliance

The financial report complies with Australian Accounting Standards (AAS). Compliance with AAS ensures that the financial report, comprising the financial statements and notes thereto, complies with International Financial Reporting Standards (IFRS). For the purpose of preparing the financial report, the Company is a not-for-profit entity.

The Directors have authorised the issue of this financial report on 12 August 2020.

(c) Revenue recognition

Revenue is measured based on the consideration to which the Company expects to be entitled in a contract with a customer and excludes amounts collected on behalf of third parties. The Company recognises revenue when it transfers control of a product or service to a customer.

Cricket revenue

Cricket revenue comprises of funding received from Cricket Australia, Sale of goods, ticketing income, membership income and sponsorship income.

Revenue from the sales of goods, ticketing income, provision of services or rental income to which the short-term lease exemption applies is recognised when the control of goods (or services) passes to the customer which is at the time that the goods are physically transferred, or the services are consumed. None of the items sold have any warranty attached to them.

Revenue related to funding from Cricket Australia are recognised in the period for which it relates and brought to account when the funds are received. Other grants are recognised as revenue when there is reasonable assurance that the Company will comply with the conditions attached to them.

Sponsorship revenue is recognised when (or as) the performance obligations specified in the agreements are satisfied.

Membership

Membership revenue is dervied through renewals and subscriptions fees. Membership revenue is recognised over time in relation to the period in which they specifically relate.

Function-related revenue

Function-related revenue relates to the sales of goods, provision of services or rental income to which the short-term lease exemption applies. Function-related revenue is recognised when the control of goods (or services) passes to the customer which is at the time that the goods are physically transferred, or the services are consumed. None of the items sold have any warranty attached to them.

(d) Finance costs

Finance costs include interest, amortisation of discounts or premiums relating to borrowings, amortisation of ancillary costs incurred in connection with arrangement of borrowings, including lease finance charges.

Finance costs are expensed as incurred.

(e) Goods and services tax

Revenues, expenses and assets are recognised net of the amount of GST, except where the amount of GST incurred is not recoverable from the ATO. In these circumstances, the GST is recognised as part of the cost of acquisition of the asset or as part of an item of the expense.

Receivables and payables are stated with the amount of GST included.

The net amount of GST recoverable from, or payable to, the ATO is included as a current asset or liability in the Statement of Financial Position.

Cash flows are included in the cash flow statement net of GST.

(f) Taxation

The Company is exempt from the payment of income tax under the Income Tax Assessment Act.

(g) Currency

Items included in the financial statements of the Company are measured using the currency of the primary economic environment in which the entity operates. The financial statements are presented in Australian dollars, which is Western Australian Cricket Association's presentation currency.

(h) Acquisitions of assets

All assets acquired including property, plant and equipment are initially recorded at their cost of acquisition at the date of acquisition, being the fair value of the consideration provided plus incidental costs directly attributable to the acquisition.

The costs of assets constructed include the cost of materials and direct labour. Directly attributable overheads and other incidental costs are also capitalised to the asset.

Assets under construction are classified accordingly until completed.

(i) Cash assets and bank overdrafts

Cash assets and bank overdrafts are carried at face value of the amounts deposited or drawn.

(j) Receivables

Trade receivables

Trade debtors are generally settled within 30 days and are therefore carried at amounts due.

Based on management's assessment, management conclude that there is no impact on the adoption of AASB 9 Financial Instruments.

(k) Inventories

Inventory is carried at the lower of cost and net realisable value.

Net realisable value

Net realisable value is determined on the basis of each inventory line's estimated selling price.

(I) Leases

The Company assesses whether a contract is or contains a lease, at inception of the contract. The Company recognises a right-of-use asset and a corresponding lease liability with respect to all lease arrangements in which it is the lessee, except for short-term leases (defined as leases with a lease term of 12 months or less) and leases of low value assets (such as tablets and personal computers, small items of office furniture and telephones). For these leases, the Company recognises the lease payments as an operating expense on a straight-line basis over the term of the lease unless another systematic basis is more representative of the time pattern in which economic benefits from the leased assets are consumed.

The lease liability is initially measured at the present value of the lease payments that are not paid at the commencement date, discounted by using the rate implicit in the lease. If this rate cannot be readily determined, the Company uses its incremental borrowing rate.

Lease payments included in the measurement of the lease liability comprise:

- · Fixed lease payments (including in-substance fixed payments), less any lease incentives receivable.
- Variable lease payments that depend on an index or rate, initially measured using the index or rate at the commencement date.
- The amount expected to be payable by the lessee under residual value guarantees.
- The exercise price of purchase options, if the lessee is reasonably certain to exercise the options.
- Payments of penalties for terminating the lease, if the lease term reflects the exercise of an option to terminate the lease.

The lease liability is presented as a separate line in the statement of financial position.

The lease liability is subsequently measured by increasing the carrying amount to reflect interest on the lease liability (using the effective interest method) and by reducing the carrying amount to reflect the lease payments made.

The right-of-use assets comprise the initial measurement of the corresponding lease liability, lease payments made at or before the commencement day, less any lease incentives received and any initial direct costs. They are subsequently measured at cost less accumulated depreciation and impairment losses.

Right-of-use assets are depreciated over the shorter period of lease term and useful life of the underlying asset

The right-of-use assets are presented as part of the Property, Plant and Equipment note in the statement of financial position.

The Company applies AASB 136 Impairment of Assets to determine whether a right-of-use asset is impaired and accounts for any identified impairment loss as described in the 'Property, plant and equipment' policy.

(m) Recoverable amount of non-current assets valued on cost basis

The carrying amounts of non-current assets valued on the cost basis are reviewed to determine whether they are in excess of their recoverable amount at balance date. If the carrying amount of a non-current asset exceeds its recoverable amount, the asset is written down to the lower amount. The write-down is recognised as an expense in the Statement of Profit or Loss and Other Comprehensive Income in the reporting period in which it occurs.

Where a group of assets working together supports the generation of cash inflows, recoverable amount is assessed in relation to that group of assets.

(n) Depreciation

Useful lives

All non-current assets have limited useful lives and are depreciated using the straight line method over their estimated useful lives.

Assets are depreciated from the date of acquisition or, in respect of internally constructed assets, from the time an asset is completed and held ready for use.

Depreciation rates and methods are reviewed annually for appropriateness. When changes are made, adjustments are reflected prospectively in current and future periods only. Depreciation is expensed.

The depreciation rates used for each class of asset are as follows:

	2020	2019
Land and Buildings	2.5% - 10%	2.5% - 10%
Plant and Equipment	10% - 33%	10% - 33%
IT Equipment	33%	-

(o) Payables

Liabilities are recognised for amounts to be paid in the future for goods or services received. Trade accounts payable are normally settled within 30 days.

(p) Provisions

Provisions are recognised when the Company has a present obligation (legal or constructive) as a result of a past event, it is probable that the Company will be required to settle the obligation, and a reliable estimate can be made of the amount of the obligation.

The amount recognised as a provision is the best estimate of the consideration required to settle the present obligation at the end of the reporting period, taking into account the risks and uncertainties surrounding the obligation. When a provision is measured using the cash flows estimated to settle the present obligation, its carrying amount is the present value of those cash flows (where the effect of the time value of money is material).

When some or all of the economic benefits required to settle a provision are expected to be recovered from a third party, a receivable is recognised as an asset if it is virtually certain that reimbursement will be received and the amount of the receivable can be measured reliably.

(q) Employee entitlements

Wages, salaries and annual leave

Liabilities for employee benefits for wages, salaries (including non-monetary benefits) and annual leave to be settled within 12 months of the reporting date represent present obligations resulting from employees' services provided up to the reporting date, calculated at undiscounted amounts based on remuneration rates that the Company expects to pay including related on-costs.

Long service leave

The provision for employee benefits for long service leave represents the present value of the estimated future cash outflows to be made resulting from employees' services provided up to reporting date.

Superannuation plan

The Company contributes to defined contribution employee superannuation plans. Contributions are charged against income as they are accrued.

(r) Financial instruments

Net fair value of financial assets and liabilities

Monetary financial assets and financial liabilities not readily traded in an organised financial market are determined by valuing them at the present value of the contractual future cash flows on amounts due from customers (reduced for expected credit losses) or due to suppliers. The carrying amounts of bank term deposits, trade debtors, term debtors, other debtors, bank overdrafts, accounts payable, bank bills and employee benefits approximate net fair value.

(s) Legal status

The Company is a not-for profit entity, incorporated under the Company's Incorporation Act 2015. The Company was converted to a Public Company Limited by Guarantee on 5 July 2018.

(t) Principal place of business

The Company's principal place of business and registered address is the WACA Ground, Nelson Crescent, East Perth, Western Australia.

			2020 (\$)	2019 (\$)
2.	Rev	venue & Other Income		
	(i)	Revenue Cricket revenue Members subscriptions	30,913,619 6,118,128	29,615,183 6,995,079
		Function-related revenue	1,316,582 38,348,328	575,936 37,186,198
	(ii)	Other Income Interest Net gain on disposal of non-current assets	219,036 -	381,035 9,949
			219,036	390,983
	Tota	I Revenue & Other Income	38,567,364	37,577,181

		2020 (\$)	2019 (\$)
3.	Surplus / (Deficit)		
	Surplus / (Deficit) has been arrived at after charging the following items:		
	Net expense including: Employee benefits, including player payments (excluding superannuation) Superannuation contributions Operating lease payments Movement in provision for employee entitlements	18,567,496 1,443,703 - (301,248)	17,739,900 1,394,766 11,508 217,312
	Depreciation: Buildings Plant and equipment and motor vehicles Computer equipment Right of Use assets	110,591 224,814 3,974 58,221	86,460 221,980 - -
		397,600	308,440
4.	Auditors remuneration		
	Audit services		
	Audit of the financial report	38,754	36,754
	Other services Other audit services	4,326	4,326
	other duale services	43,080	41,080
5.	Cash and cash equivalents		
	Cash at bank and on hand	15,465,113	2,811,715
	Short-term deposits	-	13,000,000
	Cash and cash equivalents	15,465,113	15,811,715
	The weighted average interest rate on cash assets at 30 June 2020 is 1.04% (2019: 2.15%).	15,105,115	10,01,710
	Term Deposits Maturity Timing: - 90 days - 120 days	-	7,000,000 6,000,000
		-	13,000,000
6.	Trade and other receivables		
	Current		
	Trade receivables	128,932	1,360,497
	Other receivables	101,100	65,800
		230,032	1,426,297
	The ageing of trade receivables is as follows: - Current - 30 days - 60 days - 90 days	128,152 - - - 779	1,202,353 15,637 93,485 49,022
		128,932	1,360,497
	All debtors are considered to be recoverable.		

	2020 (\$)	2019 (\$)
Inventories		
Current		
	21 500	10 770
Merchandise – at cost	21,509	12,372
Other current assets		
Prepayments	-	70,152
Property, plant and equipment		
Land and buildings		
At cost	43,211,125	42,732,738
Accumulated depreciation and accelerated depreciation	(42,490,223)	(42,379,632)
Plant and equipment and motor vehicles	720,902	353,106
At cost	7,789,954	7,736,409
Accumulated depreciation and accelerated depreciation	(7,179,095)	(6,954,282)
, team diated depreciation and decertated depreciation		
Computer equipment	610,859	782,127
At cost	884,131	870,361
Accumulated depreciation and accelerated depreciation	(874,334)	(870,361)
necessalities depreciation and deceses depreciation	9,797	-
Right of Use	, ,	
At cost	523,989	-
Accumulated depreciation and accelerated depreciation	(58,221)	-
	465,768	-
Total property, plant and equipment net book value	1,807,326	1,135,233
Total property, plant and equipment net book value	1,007,020	1,100,200
An independent valuation of a portion of the Company's land and buildings, the subject of the previously proposed ground redevelopment was carried out as at 28 April 2014 by Herron Todd White (WA) Pty Ltd, Property Consultants. The total land area of the five separate portions subject to the valuation is 24,270 sqm, as compared to a total land area of 56,656 sqm. The basis of the valuation is the unencumbered estate in fee simple in vacant possession. The independent valuation of the total land area of the five separate portions, the subject of the previously proposed ground redevelopment, is \$35.9 million. This valuation has not been reflected in the financial report. The land and buildings are recorded at cost less accumulated depreciation and Accelerated Depreciation.		
Reconciliations		
Reconciliations of the carrying amounts for each class of		
property, plant and equipment are set out below:		
Land and buildings		
Carrying amount at beginning of year	353,106	439,566
Additions	478,388	-
Depreciation	(110,591)	(86,460)
Carrying amount at end of year	720,902	353,106
Plant and equipment and motor vehicles		
Carrying amount at beginning of year	782,127	736,888
Additions	53,545	268,179
Disposals	-	(961)
	(224,814)	(221.000)
Depreciation	(224,014)	(221,980)

	2020 (\$)	2019 (\$)
9. (continued)		
Computer equipment		
Carrying amount at the beginning of year	-	-
Additions	13,770	-
Depreciation	(3,974)	
Carrying amount at end of year	9,797	
Right of Use		
Carrying amount at the beginning of year	-	-
Additions Depreciation	523,989 (58,221)	-
Carrying amount at end of year	465,768	
Total property, plant and equipment net book value	1,807,326	1,135,233
10. Payables		
Current		
Trade creditors and accruals	4,577,317	6,130,612
11. Lease Liabilities		
Current	62,916	-
Non-current	408,861	-
	471,777	-
Maturity analysis		
Year 1	62,916	-
Year 2	61,262	-
Year 3	59,652	-
Year 4 Year 5	58,083 56,556	-
Onward	173,307	-
	471,777	-
12. Provisions		
Current		
Employee leave entitlements	1,022,568	1,231,283
Other Employee Related	64,700	-
Payroll Tax	192,560	-
	1,279,828	1,231,283
Non-current Employee leave entitlements	200,973	293,505
Employee leave entitiements	200,973	293,505
	200,373	233,303
Number of employees	No.	No.
Number of permanent employees at year end	97	109
(Full Time Equivalent, "FTE")		

			NOTES	2020 (\$)	2019 (\$)
13.	Def	erred Income			
		ent (i)		225,831	282,703
	Non-	current		_	_
	(i) D	eferred income (current) relates to government grants, corporate ospitality and other income received which relates to the year nded 30 June 2021.			
14.	Commitments				
	Operating lease payable commitments				
	Future commitments are payable:				
	Within one year			-	63,466
	One year or later and no later than five years			-	266,505
	Lat	er than five years		-	293,825
				-	623,795
	Fees	lisclosure of future commitments related to Venue Signage Rental payable to Optus Stadium has been replaced with AASB 16. Please to note 1 for detailed disclosure.			
15.	Not	es to the statement of cash flows			
	(i)	Reconciliation of cash			
		For the purposes of the cash flow statement, cash includes cash on hand and at bank and short-term deposits at call, net of outstanding bank overdrafts. Cash at the end of the financial year as shown in the cash flow statement is reconciled to the related items in the Statement of Financial Position as follows:			
		Cash assets	5	15,465,113	15,811,715
	(ii)	Reconciliation of operating (deficit) / surplus to net cash provided by operating activities			
		Operating surplus / (deficit)		250,588	(347,016)
		Add/(less) items classified as investing/financing activities: (Gain)/loss on sale of non-current assets		-	(9,949)
		Add/(less) non-cash items:			
		Interest payment on Lease		12,403	-
		Depreciation		397,600	308,440
		Net cash provided by / (used in) operating activities before change in assets and liabilities		660,592	(48,524)
		Change in assets and liabilities during the financial year: (Increase)/decrease in receivables (Increase)/decrease in inventories (Increase)/decrease in other current assets Increase/(decrease) in payables Increase/(decrease) in provisions Increase/(decrease) in deferred income		1,196,265 (9,137) 70,152 (1,553,295) (43,988) (56,873)	(971,378) (724) (69,322) 3,019,109 217,313 (112,669)
		Net cash provided by operating activities		263,716	2,033,804
	(iii)	Non cash investing and financing activities During the financial year or previous year, the Company did not acquire any property, plant and equipment by means of financing leases.			

			2020 (\$)	2019 (\$)
16.	Related parties			
	(i)	Board Members		
		No member of the Board has received any remuneration from the Company and no member of the Board has entered into a contract with the Company since the end of the previous financial year.		
	(ii)	Key management personnel compensation		
		Key management personnel compensation is set out below. The key management personnel are the executives with the authority for the strategic direction and management of the Company.		
		Short-term benefit	1,550,888	1,584,695
		Post Employment benefit	141,054	140,903
		Other Long-Term benefit	(14,187)	(3,322)
			1,677,755	1,722,276

17. Segment reporting

The Company operates wholly within one business segment, being the promotion and development of cricket in Western Australia.

18. Events subsequent to balance date

The potential impact to revenue due to COVID-19 in coming years are:

- Reduced Cricket Australia funding by up to 25%;
- The potential changing cricket fixtures (for both International matches and the Big Bash League) due to COVID-19 Government restrictions;
- Estimated lower (with the potential of no) crowd attendance;
- · Potential reduction in membership and sponsorship revenue if cricket content is compromised.

In August 2020, the State Government announced it would match the \$30,000,000 Federal Government funding commitment towards the WACA Ground Improvement Project. The details of the terms of the State funding are being finalised. The WACA is also currently exploring the feasibility of an opportunity to partner with the City of Perth to include an aquatic facility at the WACA Ground.

Other than the impact of COVID-19, there has not arisen any other item, transaction or event of a material nature likely, in the opinion of the members of the Board, to affect materially the operations or state of affairs of the Company in future financial years.

19. Contingent Liabilities

- a. A development agreement ("Agreement") signed on 1 July 2009 with Ripple Holdings Pty Ltd, as trustee for the WACA Redevelopment Trust ("Ripple") and Ascot Capital Limited ("Ascot") was terminated on 16 December 2013. A liability not included in these Financial Statements may arise if any of the land which is the subject of the Agreement ("development land") is alienated in any way, including sale, assignment, transfer or lease.
 - If any of the development land is alienated, the Company must pay to Ripple an amount equal to 8% of the increase in value of the development land and which increased value is attributable to any development approvals obtained by Ripple on behalf of the Company, less preliminary costs incurred, ("Value Uplift"). No amount is payable if the Value Uplift is less than the preliminary costs. The total of preliminary costs incurred over the duration of the project was \$7.0m. The Value Uplift fee would be to a maximum of the relevant proceeds received up to the relevant date. The contingent liability in relation to the Value Uplift fee is dependent on uncertain future events and is not quantifiable at this stage. Although there is a contingent liability it is expected that any amount payable will be met with funds generated from the sale, assignment, transfer or lease of the development land should this event eventuate.
- b. The Company was fined \$150,000 by Cricket Australia on 4 November 2016 for an administrative breach relating to player payments for the 2015/16 Big Bash League season. Of the total fine, \$145,000 is suspended for five years to 4 November 2021, when it will be waived if no similar offence has been committed.

20. Fair Value of Financial Instruments

The carrying amounts of cash and cash equivalents, trade and other receivables, trade and other payables are a reasonable approximation of their fair values, on account of their short maturity cycle.

The fair value of interest-bearing liabilities are estimated to approximate the carrying value as; the finance leases have a short maturity cycle; the bank bill facility is subject to variable 90 day bill rates which approximate to the long term bond rate.

21. Risk Management Strategies

The Company is primarily exposed to credit risks, liquidity risks and interest rate risks.

Credit Risks: Credit terms are provided to certain sponsorships' clients; cricket clubs and associations in respect of goods and services provided. In relation to sponsorships' clients, the credit risk is assessed as part of contract negotiation and payment terms are stipulated in contracts. In relation to cricket clubs and associations the credit risk is mitigated to a large extent by the ability to offset amounts due from payments made. Aged debtor reports are produced on a monthly basis and overdue accounts are followed up.

Liquidity Risks: Cash is required to meet ongoing expenses including wages, taxes and creditors. Cash flows are forecast, monitored and planned on a weekly basis so as to ensure cash is available as required. Funds are invested on such terms as to ensure sufficient liquidity.

Interest Rate Risks: Movements in interest rates are monitored on a monthly basis.

There are presently no currency or other pricing risks facing the Company.

22. Investment in Subsidiary Company

Perth Scorchers Pty Ltd ("company"), a proprietary company limited by shares was registered on 27 April 2011. The issued share capital of the company is 100 ordinary shares of \$1 each. The shares are wholly owned by the Company. The company is dormant.

As at 30 June 2020 other than \$100 share capital, the company had no assets or liabilities. The investment in the company has not been brought to account in the accounts of the Company. Consolidated financial statements have not been prepared at 30 June 2020 due to immateriality.

Board's Declaration

The Board Directors declare that:

- a) in the Board's opinion, there are reasonable grounds to believe that the Company will be able to pay its debts as and when they become due and payable;
- b) in the Board's opinion, the attached financial statements are in compliance with International Financial Reporting Standards, as stated in note 1(b) to the financial statements; and
- c) in the Board's opinion, the attached financial statements and notes thereto are in accordance with the Company Rules, including compliance with accounting standards and giving a true and fair view of the financial position and performance of the Company.

Signed in accordance with a resolution of the Board Directors:

Dated at Perth this 12th day of August 2020.

On behalf of the Board

The Hon TK Waldron | Chair

DJ Bailey | Chair, Audit & Risk Committee

Deloitte Touche Tohmatsu ABN 74 490 121 060

Tower 2 Brookfield Place 123 St Georges Terrace Perth WA 6000 GPO Box A46 Perth WA 6837 Australia

Tel: +61 8 9365 7000 Fax: +61 8 9365 7001 www.deloitte.com.au

Independent Auditor's Report to the members of Western Australian Cricket Association Limited

Opinion

We have audited the financial report of Western Australia Cricket Association Limited (the "Company") which comprises the statement of financial position as at 30 June 2020, the statement of profit or loss and other comprehensive income, the statement of changes in accumulated funds and the statement of cash flows for the year then ended, and notes to the financial statements, including a summary of significant accounting policies, and the directors' declaration.

In our opinion, the accompanying financial report of the Company is in accordance with the *Corporations Act 2001*, including:

- (i) giving a true and fair view of the Company's financial position as at 30 June 2020 and of its financial performance for the year then ended; and
- (ii) complying with Australian Accounting Standards and the Corporations Regulations 2001.

Basis for Opinion

We conducted our audit in accordance with Australian Auditing Standards. Our responsibilities under those standards are further described in the *Auditor's Responsibilities for the Audit of the Financial Report* section of our report. We are independent of the Company in accordance with the auditor independence requirements of the *Corporations Act 2001* and the ethical requirements of the Accounting Professional and Ethical Standards Board's APES 110 *Code of Ethics for Professional Accountants* (including Independence Standards) (the Code) that are relevant to our audit of the financial report in Australia. We have also fulfilled our other ethical responsibilities in accordance with the Code.

We confirm that the independence declaration required by the *Corporations Act 2001*, which has been given to the directors of the Company, would be in the same terms if given to the directors as at the time of this auditor's report.

We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our opinion.

Other Information

The directors are responsible for the other information. The other information comprises the information included in the Company's annual report for the year ended 30 June 2020, but does not include the financial report and our auditor's report thereon.

Liability limited by a scheme approved under Professional Standards Legislation.

Member of Deloitte Asia Pacific Limited and the Deloitte Network.

Our opinion on the financial report does not cover the other information and we do not express any form of assurance conclusion thereon.

In connection with our audit of the financial report, our responsibility is to read the other information and, in doing so, consider whether the other information is materially inconsistent with the financial report or our knowledge obtained in the audit, or otherwise appears to be materially misstated. If, based on the work we have performed, we conclude that there is a material misstatement of this other information, we are required to report that fact. We have nothing to report in this regard.

Responsibilities of the Directors for the Financial Report

The directors of the Company are responsible for the preparation of the financial report that gives a true and fair view in accordance with Australian Accounting Standards and the Corporations Act 2001 and for such internal control as the directors determine is necessary to enable the preparation of the financial report that gives a true and fair view and is free from material misstatement, whether due to fraud or error.

In preparing the financial report, the directors are responsible for assessing the ability of the Company to continue as a going concern, disclosing, as applicable, matters related to going concern and using the going concern basis of accounting unless the directors either intend to liquidate the Company or to cease operations, or has no realistic alternative but to do so.

Auditor's Responsibilities for the Audit of the Financial Report

Our objectives are to obtain reasonable assurance about whether the financial report as a whole is free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not a guarantee that an audit conducted in accordance with the Australian Auditing Standards will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of this financial report.

As part of an audit in accordance with the Australian Auditing Standards, we exercise professional judgement and maintain professional scepticism throughout the audit. We also:

- Identify and assess the risks of material misstatement of the financial report, whether due to
 fraud or error, design and perform audit procedures responsive to those risks, and obtain
 audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk
 of not detecting a material misstatement resulting from fraud is higher than for one resulting
 from error, as fraud may involve collusion, forgery, intentional omissions,
 misrepresentations, or the override of internal control.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control.

- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by the directors.
- Conclude on the appropriateness of the directors' use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to events or conditions that may cast significant doubt on the Company's ability to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the financial report or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Company to cease to continue as a going concern.
- Evaluate the overall presentation, structure and content of the financial report, including the disclosures, and whether the financial report represents the underlying transactions and events in a manner that achieves fair presentation.

We communicate with the directors regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

DELOUTE TOUCHE TOUMATSU

DELOITTE TOUCHE TOHMATSU

John SibenalerPartner

Chartered Accountants Perth, 13 August 2020

The Board of Directors Western Australian Cricket Association Limited WACA Ground, Nelson Crescent East Perth, WA 6004

13 August 2020

Dear Board Members

Deloitte Touche Tohmatsu ABN 74 490 121 060

Tower 2 Brookfield Place 123 St Georges Terrace Perth WA 6000 GPO Box A46 Perth WA 6837 Australia

Tel: +61 8 9365 7000 Fax: +61 8 9365 7001 www.deloitte.com.au

Western Australian Cricket Association Limited

In accordance with section 307C of the *Corporations Act 2001*, I am pleased to provide the following declaration of independence to the directors of Western Australian Cricket Association Limited.

As lead audit partner for the audit of the financial report of Western Australian Cricket Association Limited for the year ended 30 June 2020, I declare that to the best of my knowledge and belief, there have been no contraventions of:

- (i) the auditor independence requirements of the Corporations Act 2001 in relation to the audit; and
- (ii) any applicable code of professional conduct in relation to the audit.

Yours sincerely

DELOUTE TOUCHE TOUMATSU

DELOITTE TOUCHE TOHMATSU

John Sibenaler Partner

Chartered Accountants

Liability limited by a scheme approved under Professional Standards Legislation. Member of Deloitte Asia Pacific Limited and the Deloitte Network.

Constitution of the Western Australian Cricket Association Limited

5 July 2018



Table of Contents

KEY ORGANISATIONAL INFORMATION

- 1.1 Definitions and Interpretation
- 1.2 Name and nature of the Organisation
- 1.3 Interaction with Act
- 1.4 Objects
- 1.5 No distribution to Members
- 1.6 Recognition of heritage

2. MEMBERS

- 2.1 Classes of membership
- 2.2 Applications
- 2.3 No transfers
- 2.4 Ceasing to be a Member
- 2.5 Resignation
- 2.6 Expulsion or suspension
- 2.7 Variation of Classes and Class rights
- 2.8 Certificates or Membership Cards
- 2.9 Suspension of Member rights for non-payment of fees

3. FEES

- 3.1 Fees
- 3.2 Interest

4. PROCEEDINGS OF MEMBERS

- 4.1 Who can call meetings of Members
- 4.2 Annual General Meeting
- 4.3 How to call meetings of Members
- 4.4 Right to attend, speak and vote at Member meetings
- 4.5 Meeting at more than one place
- 4.6 Quorum
- 4.7 Chair
- 4.8 General conduct of meetings
- 4.9 Resolutions of Members
- 4.10 Polls
- 4.11 Adjourned, cancelled and postponed meetings
- 4.12 Number of votes
- 4.13 Objections to qualification to vote
- 4.14 Proxies and powers of attorney

5. DIRECTORS

- 5.1 Number and Qualification of Directors
- 5.2 Cricket Appointed Directors
- 5.3 Appointment of Member Elected Directors by Members
- 5.4 Appointment of Board Appointed Directors
- 5.5 Retirement and re-appointment of Directors

- 5.6 Maximum term for a Director
- 5.7 Vacation of office of Director
- 5.8 No Alternate Directors
- 5.9 Payment to Directors prohibited
- 5.10 Skills, experience and attributes of Directors
- 5.11 Consents and Undertakings

6. OFFICERS AND OFFICEHOLDERS

- 6.1 Patron-in-Chief
- 6.2 Patron(s)
- 6.3 Chair and Deputy Chair
- 6.4 Chief Executive Officer
- 6.5 Secretary(s)
- 6.6 Indemnity and insurance

POWERS OF THE ORGANISATION AND BOARD

- 7.1 General powers
- 7.2 Execution of Documents
- 7.3 Committees and delegates
- 7.4 Attorney or agent
- 7.5 Affiliated Organisations7.6 Duties of Directors
- 7.7 Conflicts of Interest
- 7.8 Cricket Australia

8. PROCEEDINGS OF BOARD

- 8.1 Written resolutions of the Directors
- 8.2 Meetings of the Board
- 8.3 Who can call meetings of the Board
- 8.4 How to call meetings of the Board
- 8.5 Quorum
- 8.6 Resolutions of the Board

9. BOARD AND OTHER COMMITTEES

- 9.1 Board may establish Committees
- 9.2 Terms of Reference and Governance
- 9.3 Review
- 9.4 SGDC

10. BY-LAWS

- 10.1 By-laws
- 10.2 By-law Matters
- 10.3 No conflict with Constitution
- 10.4 Making of By-laws
- 10.5 Force and Effect of By-Laws

11. RECORDS

- 11.1 Financial Year
- 11.2 Accounts
- 11.3 Books, accounts and records
- 11.4 Audit
- 11.5 Minutes

12. ALTERATIONS TO CONSTITUTION

13. NOTICES

- 13.1 Notice to Members
- 13.2 Notice to Directors
- 13.3 Notice to the Organisation
- 13.4 Time of service
- 13.5 Signatures

14. WINDING UP

- 14.1 Transfer of surplus
- 14.2 Application to Court

15. TRANSITIONAL PROVISIONS

- 15.1 Transitional Arrangements
- 15.2 Transitional Provisions Paramount
- 15.3 Continuance of Transitional Provisions

SCHEDULE 1

Definitions and interpretation

SCHEDULE 2

Key Information

SCHEDULE 3

Transitional Provisions

SCHEDULE 4

Western Australian Cricket Association Limited, By-Laws

1 Key Organisational Information

1.1 Definitions and Interpretation

Schedule 1 applies and forms part of this Constitution to assist in its interpretation.

1.2 Name and nature of the Organisation

- (a) The name of the Organisation is Western Australian Cricket Association Limited.
- (b) The Organisation is a public company limited by guarantee under the Act.
- (c) Each Member undertakes to contribute an amount not exceeding ONE dollar (AUD\$1.00) to the property of the Organisation if the Organisation is wound up:
 - (i) at a time when that person is a Member; or
 - (ii) within one year of the time that person ceased to be a Member,

for or towards:

- (iii) payment of the debts and liabilities of the Organisation contracted before that person ceased to be a Member;
- (iv) payment of the costs, charges and expenses of winding up the Organisation; and
- adjustment of the rights of the contributories among Members.

1.3 Interaction with Act

The replaceable rules in the Act do not apply to the Organisation.

1.4 Objects

- (a) The objects of the Organisation are:
 - to lead, promote, nurture, develop, co-ordinate and administer, and to provide governance leadership for, cricket at all levels of the sport including but not limited to representative cricket, premier cricket, junior cricket and community based participation in the sport of cricket, both active and passive, including through affiliation and co operation with other organisations;
 - (ii) to maintain membership and to actively engage with and exercise its membership rights (including rights to board representative appointment or nomination, as applicable) of Cricket Australia, and to maintain and promote relations with other member organisations of Cricket Australia;
 - (iii) to promote, develop and manage, or otherwise deal with the WACA Ground as a strategic asset of the Organisation ancillary to and in furtherance of these objects;
 - (iv) to provide facilities, amenities and services for the benefit of Members consistent with these objects but subject nevertheless to Article 1.5(a)

(b) The Organisation:

- (i) will only apply the income and property of the Organisation in promoting the objects of the Organisation; and
- (ii) must not transfer its assets and engagements to, or amalgamate with any corporation, association or organisation which does not, to the same extent as this Constitution:
 - (A) restrict the application of its income and property and prohibit the making of distributions to its Members; and
 - (B) have objects similar to the objects of the Organisation under this Article.

1.5 No distribution to Members

- (a) Subject to Article 1.5(b), the Organisation must not make any payment or distribution or apply any part of the income or property of the Organisation, to any Member, including by way of dividend, surplus on winding up or otherwise.
- (b) Article 1.5(a) does not prevent the Organisation, with the approval of the Board acting in good faith (including through delegation of power to the CEO and to management), paying:
 - reasonable remuneration to a Member who is an employee of the Organisation;
 - subject to Article 5.9, reasonable remuneration to a Member who is an officer of the Organisation;
 - (iii) reasonable remuneration in consideration for services rendered or goods supplied by a Member to the Organisation in the ordinary course of business;
 - (iv) interest, at a reasonable rate based on commercial "arm's length" terms, on money borrowed by the Organisation from a Member;
 - (v) reasonable rent based on commercial "arm's length" terms, for premises leased to the Organisation by a Member:
 - (vi) out-of-pocket expenses properly and reasonably incurred by a Member at the request of the Organisation and for, or on behalf of, the Organisation;
 - (vii) out of pocket expenses properly and reasonably incurred by a Member in his/her capacity as a Director, in connection with the affairs of the Organisation; or
 - (viii) any other reasonable amount of similar character to those described in this Article 1.5(b).

1.6 Recognition of heritage

The instrumental role of premier cricket (formerly known as district cricket) stakeholders in Western Australia in the foundation of the Organisation is acknowledged and respected.

2 Members

2.1 Classes of membership

The:

- (a) various Classes of Membership and status within a Class;
- (b) restrictions (if any) in the number of Members or the number of Members within each Class or status within a Class:
- (c) qualifications for admission to each Class or status within a Class; and
- (d) rights attached to being a Member in each Class, are as prescribed in the By-laws from time to time.

2.2 Applications

- (a) Subject to Article 2.1 and the By-laws, a person may apply for Membership in the manner prescribed in this Article.
- (b) An applicant for Membership must:
 - sign and deliver to the Organisation an application in the Applicable Form including specification as to the Class of Membership being applied for; and
 - (ii) pay any Fees with respect to that application,

as specified in the By-laws.

(c) The Board, including on recommendation of management or a committee of the Board constituted with terms of reference including such purpose, may determine whether an applicant may become a Member.

- (d) The Board is not required to give any reason for the approval or rejection of any application to become a Member.
- (e) If an application to become a Member is accepted, the Organisation must:
 - give written notice of the acceptance to the applicant including details of the Class of Membership and the rights that are then attached to that Class;
 - (ii) request payment of any outstanding amount owing for any Fees incidental to such Membership; and
 - (iii) upon payment in full of Fees, enter the applicant's name in the Register as a Member of the relevant Class.
- (f) If an application to become a Member is rejected, the Organisation must:
 - give written notice of the rejection to the applicant; and
 - (ii) refund in full any Fees, in addition to any Membership nomination fee (which shall not be refundable), paid to the Organisation by the applicant when applying for Membership.
- (g) The Board may, at its discretion, determine to close offering Membership, either generally or with respect to any particular Class or Classes, to new applicants, in which case the Organisation may keep the names and particulars of any new applicants on a waiting list, but without assurance of Membership becoming available within any timeframe or priority being given to those on such a waiting list.

2.3 No transfers

Membership, and the rights of being a Member, are personal to the Member and are not transferable whether by operation of law or otherwise.

2.4 Ceasing to be a Member

- (a) A person will cease to be a Member if:
 - (i) that person resigns in accordance with Article 2.5;
 - (ii) that person is expelled under Article 2.6; or
 - (iii) a Cessation Event occurs in respect of that person.
- (b) The estate of a deceased Member is not released from any accrued liability prior to their death in respect of that person being a Member.

2.5 Resignation

- (a) A member may resign as a Member by giving the Organisation notice in writing.
- (b) Unless the notice provides otherwise, a resignation by a Member takes effect immediately on the giving of that notice to the Organisation.
- (c) A Member is not released from any accrued liability in respect of their Membership merely by giving notice of resignation.

2.6 Expulsion or suspension

- (a) Subject to this Article 2.6, the Board may resolve to:
 - (i) expel a Member; or
 - (ii) suspend a Member:
 - (A) for such period; and
 - (B) from enjoying such rights and privileges of Membership, as the Board may determine and give written notice of to the Member;
- (b) The Board may only act under Article 2.6(a)(i) if:
 - an Expulsion Event occurs in respect of the Member; and

- (ii) other than in the case of non-payment by the Member of Fees or other moneys owing to the Organisation, the Organisation gives that Member at least 10 Business Days' notice in writing:
 - (A) stating the Expulsion Event and that the Member is liable to be expelled;

and

- (B) informing the Member of the Member's right under Article 2.6(c)(i).
- (c) Other than in the case of non-payment by the Member of Fees or other moneys owing to the Organisation, before passing any resolution under Article 2.6(a)(i), the Board:
 - must allow the Member to give to the Board, either orally or in writing, any explanation or defence of the Expulsion Event: and
 - (ii) may adopt other procedures to aid the resolution of any complaint against the Member, including the appointment of complaints committees, conciliators and mediators if and as the Board may determine is appropriate in all the circumstances of the matter.
- (d) Where a resolution is passed under Article 2.6(a), the Organisation must give the Member, notice ("Discipline Notice") in writing of the expulsion or suspension, within 10 Business Days of the resolution.
- (e) Other than in the case of non-payment by the Member of Fees or other moneys owing to the Organisation, a member may, by notice in writing to the Organisation within 10 Business Days of issue of a Discipline Notice, request that a resolution to negate that expulsion (but not suspension) of that Member under Article 2.6(a) be moved at the next AGM of the Organisation.
- (f) If a request under Article 2.6(e) is made, the Board must table at the AGM of the Organisation an invitation for that meeting to move a resolution to overturn the decision of the Board to expel the Member concerned.
- (g) A Board resolution for expulsion under Article 2.6(a) is deemed to take effect:
 - (i) if the Member does not give a notice under Article 2.6(e), on the date of that Board resolution:
 - (ii) if the Member gives notice under Article 2.6(e), as an immediate suspension of Membership and Member rights, and as an expulsion of Membership effective from the conclusion of the next AGM, unless the resolve of that AGM is to overturn the resolution of the Board to expel the Member.
- (h) The Board may reinstate an expelled or suspended Member on any terms and at any time as the Board determines, including as a pre condition, a requirement that all amounts due but unpaid by the expelled or suspended Member to the Organisation are paid.
- (i) Nothing in this Article limits the effect of any By-law that may be passed to temporarily deny a Member access to any facilities under the control or management of the Organisation where that Member's behaviour has fallen short of acceptable standards as prescribed in the By-laws.

2.7 Variation of Classes and Class rights

Subject to the Act, this Constitution and the By-laws the Organisation may:

- (a) vary or cancel rights attached to being a Member of a Class; or
- (b) convert all Members from an existing Class to another Class.

2.8 Certificates or Membership Cards

(a) The Organisation may issue to each Member, free of charge, a certificate or Membership Card evidencing that person as a Member or as a Member of a particular Class.

- (b) The Organisation may issue a replacement certificate or Membership Card of being a Member if:
 - (i) the Organisation receives and cancels the Member's existing certificate or Membership Card or the Organisation is satisfied that the Member's existing certificate or Membership Card is lost or destroyed and the Organisation has received a declaration from the Member in Applicable Form attesting to that loss of destruction and indemnifying the Organisation against any loss, liability or claim arising from such loss or destruction and the issue of a replacement certificate or Membership Card; and
 - the Member pays any applicable administrative Fee as the By-laws may prescribe.
- (c) The certificate or Membership Card at all times remains the property of the Organisation and a retiring, expelled or suspended Member must promptly return their certificate or Membership Card to the Organisation and must not display or use the Membership Card to gain entitlements that are only available to Members.

2.9 Suspension of Member rights for non-payment of fees

Without limiting any other provision of this Constitution or the By-laws, and without limiting any right of recovery for unpaid Fees at law, if a Member has not paid any Fees which are past due and payable, all rights attaching to that Member's Membership are automatically suspended until payment of the outstanding Fee is paid.

3 Fees

3.1 Fees

- (a) The Organisation may require the payment of fees or levies by Members in the amounts and at the times as the By-laws prescribe.
- (b) The Organisation may make Fees payable for one or more Members, or Classes of Members, for different amounts and at different times.
- (c) Pursuant to Article 3.1(a), the Organisation may, from time to time, give notice to Members or to any particular Member or Class of Members:
 - (i) revoking or postponing Fees;
 - (ii) extending the time for payment of Fees;
 - (iii) allowing for payment of Fees by instalments or with discount; or
 - (iv) stipulating the amount, the time, the method and the place of payment of Fees.

3.2 Interest

- (a) A Member must pay to the Organisation:
 - interest at the rate prescribed in the By-laws on any Fees which are not paid on, or before, the time appointed for payment, from the time appointed for payment to the time of the actual payment; and
 - (ii) expenses incurred by the Organisation, including legal and internal administration expenses, because of the failure to pay, or late payment of, that amount.
- (b) The Organisation may waive payment of all or any part of an amount payable under Article 3.2(a).

4 Proceedings of Members

4.1 Who can call meetings of Members

(a) Subject to the Act, the Board may call a meeting of Members at a time and place as the Board resolves in accordance with Part 2G.2 of the Act.

- (b) The Board must call and arrange to hold a general meeting on the request of Members made in accordance with the Act.
- (c) The Members may call and arrange to hold a general meeting in accordance with Part 2G.2 of the Act.
- (d) Division 4 of Part 2G.2 of the Act also applies to the Organisation.

4.2 Annual General Meeting

- (a) The Organisation must hold an AGM as required by, and in accordance with the Act.
- (b) The business of an AGM may include any of the following, even if not referred to in the notice of the meeting:
 - the consideration of the annual financial report, the Directors' report and the auditor's report for the Organisation;
 - (ii) the appointment of the auditor of the Organisation upon terms recommended by the Board; and
 - (iii) the election of Directors.

4.3 How to call meetings of Members

- (a) The Organisation must give not less than the Prescribed Notice of a meeting of Members.
- (b) Notice of a meeting of Members must be given to each Member, each Director, the Company Secretary and any auditor of the Organisation in the manner set out in Article 13.
- (c) Subject to Article 4.11(h), a notice of a meeting of Members must:
 - set out the place, date and time for the meeting (and if the meeting is to be held in two or more places, the technology that will be used to facilitate this);
 - state the general nature of the business of the meeting; and
 - (iii) set out or include any other information or documents specified by the Act.
- (d) Subject to the Act, anything done (including the passing of a resolution) at a meeting of Members is not invalid merely because a person does not receive notice of the meeting or the Organisation inadvertently does not give notice of the meeting to a person or there may be a non-material irregularity in the calling of the meeting or the notice of the meeting.

4.4 Right to attend, speak and vote at Member meetings

- (a) Subject to Article 4.8 and any limitation on the rights attaching to Membership of any Class of Member, each Member, the Directors, the Company Secretary and any auditor of the Organisation (on any part of the affairs of the Organisation that concerns the auditor in his/her capacity as auditor) is entitled to attend and speak at any meeting of Members.
- (b) Subject to Article 4.8 and any limitation on the rights attaching to Membership of any Class of Members, each Member, is entitled to vote at any meeting of Members.

4.5 Meeting at more than one place

- (a) A meeting of Members may be held in two or more places linked together by any technology that:
 - gives the Members as a whole in those places a reasonable opportunity to participate in proceedings;
 - enables the chair of the meeting to be aware of proceedings in each place; and
 - (iii) enables the Members in each place to vote on a show of hands and on a poll.

- (b) If a meeting of Members is held in two or more places under Article 4.5(a):
 - a Member present at one of the places is taken to be present at the meeting; and
 - (ii) the chair of that meeting may determine at which place the meeting is taken to have been held.

4.6 Quorum

- (a) Subject to Article 4.6(e), a quorum for a meeting of Members or any Class of Members is that number of Members entitled to vote at that meeting as specified in Item 1 of Schedule 2.
- (b) In determining whether a quorum for a meeting of Members is present:
 - (i) where a person is present as a Member entitled to vote at the Meeting and as a proxy or attorney of another Member entitled to vote at the Meeting, that person is counted separately for each appointment provided that there is at least four other Members entitled to vote at the meeting present; and
 - (ii) where a person is present as a proxy or attorney for more than one Member each of whom is entitled to vote at the meeting, that person is counted separately for each appointment provided that there is at least four other Members entitled to vote at the meeting present.
- (c) A quorum for a meeting of Members must be present at the commencement of the meeting. If a quorum is present at the commencement of a meeting of Members, it is taken to be present throughout the meeting unless the chair of the meeting otherwise determines.
- (d) If a quorum is not present within 30 minutes after the time appointed for a meeting of Members:
 - (i) if the meeting was called under Article 4.1(b) or Article 4.1(c), the meeting is dissolved; and
 - (ii) for any other meeting, the meeting is adjourned to the date, time and place as the meeting chair may appoint, or failing any such appointment, to the same day in the next week at the same time and place as the meeting adjourned.
- (e) If a quorum is not present within 30 minutes after the time appointed for an adjourned meeting of Members:
 - (i) if there are not less than 50% of the number of Members present as represent a quorum under Item 1 of Schedule 2, then they shall constitute a quorum;
 - (ii) otherwise, the meeting is dissolved.

4.7 Chair

- (a) The Chair must (if present within 15 minutes after the time appointed for the holding of the meeting and willing to act) chair each meeting of Members.
- (b) If at a meeting of Members:
 - (i) there is no Chair;
 - the Chair is not present within 15 minutes after the time appointed for the holding of a meeting of Members; or
 - the Chair is present within that time but is not willing or able to chair all or part of that meeting,

then the Deputy Chair, or in default of the Deputy Chair, another Board member, or a Member, elected by majority vote of Board members present will chair all or part of the meeting of Members.

- (c) Subject to Article 4.7(a), if at a meeting of Members:
 - (i) a chair of that meeting has not been elected by the Board members under Article 4.7(b); or

the chair elected by the Board members is not willing or able to chair all or part of a meeting of Members,

then the Members present must elect a Member, present and willing to act, to chair that meeting.

4.8 General conduct of meetings

- (a) Subject to the Act and any relevant provisions in the By-laws, the chair of a meeting of Members is responsible for the general conduct of that meeting and for the procedures to be adopted at that meeting and rulings by the chair on such matters made in good faith shall be prima facie binding on the meeting and its Members.
- (b) The chair of a meeting of Members must give those entitled to attend and speak at such meetings a reasonable opportunity to ask questions and make comments at the meeting (including the auditor (if any)).
- (c) The chair of a meeting of Members may delegate any power conferred by this Article 4.8 to any person.
- (d) The powers conferred on the chair of a meeting of Members under this Article 4.8 do not limit any powers conferred at law upon a chair of a meeting.

4.9 Resolutions of Members

- (a) Subject to the Act and other than for Special Resolutions, a resolution of Members is passed if more votes are cast in favour of the resolution by Members entitled to vote on the resolution than are cast against the resolution.
- (b) Unless a poll is requested in accordance with Article 4.10, a resolution put to the vote at a meeting of Members is to be decided on a show of hands.
- (c) A declaration by the chair of a meeting of Members that a resolution has been passed on a show of hands, passed by a particular majority, or not passed, and an entry to that effect in the minutes of the meeting, is sufficient evidence of that fact, unless proved incorrect.

4.10 Polls

- (a) A poll may be demanded on any resolution at a meeting of Members except:
 - (i) the election of a chair of that meeting; or
 - (ii) the adjournment of that meeting.
- (b) A poll on a resolution at a meeting of Members may be demanded by:
 - at least five Members present and entitled to vote on that resolution; or
 - (ii) the chair of that meeting.
- (c) A poll on a resolution at a meeting of Members may be demanded:
 - (i) before a vote on that resolution is taken; or
 - before, or immediately after, the result of the vote on that resolution on a show of hands is declared.
- (d) A demand for a poll may be withdrawn
- (e) A poll demanded on a resolution at a meeting of Members must be taken in the manner and at the time and place the chair directs.
- (f) The result of a poll demanded and taken on a resolution of a meeting of Members is a resolution of that meeting.
- (g) A demand for a poll on a resolution of a meeting of Members does not prevent the continuance of that meeting or that meeting dealing with any other business pending the taking of the poll or the declaration of the result of the poll.

4.11 Adjourned, cancelled and postponed meetings

- (a) Subject to the Act, the chair of a meeting of Members:
 - (i) may; and

(ii) must, if the Members present with a majority of votes that may be cast at that meeting agree or direct the chair to do so.

adjourn a meeting of Members to any day, time and place.

- (b) No person other than the chair of a meeting of Members may adjourn that meeting.
- (c) Notice of an adjourned meeting is only required if the period of adjournment exceeds the Prescribed Period.
- (d) Only business left unfinished is to be transacted at a meeting of Members resumed after an adjournment.
- (e) Subject to the Act and this Article 4.11, the Board may at any time postpone or cancel a meeting of Members by giving notice, not less than 3 Business Days before the time at which the meeting was to be held, to each person to whom the notice of the meeting was required to be given.
- (f) A general meeting called under Article 4.1(b) must not be cancelled by the Board without the consent of the Members who requested the meeting.
- (g) A general meeting called under Article 4.1(c) must not be cancelled or postponed by the Board without the consent of the Members who called the meeting.
- (h) A notice adjourning or postponing a meeting of Members must set out the place, date and time for the revised meeting (and if the revised meeting is to be held in two or more places, the technology that will be used to facilitate this).

4.12 Number of votes

- (a) Subject to this Constitution and any rights or restrictions attached to a Class of Membership as prescribed in the By-laws, on a show of hands at a meeting of Members, every Member present has one vote, and on a poll at a meeting of Members, every Member present in person or by proxy has one vote.
- (b) In the case of an equality of votes on a resolution at a meeting of Members, the chair of that meeting has a casting vote on that resolution both on a show of hands and on a poll, in addition to any vote the chair of the meeting has in respect of that resolution.
- (c) A Member present at a meeting of Members is not entitled to vote on any resolution, whether in person or by proxy or attorney, while that Member is the subject of suspension under Article 2.6 or their Member rights are under suspension under Articles 2.9.
- (d) A Member is not entitled to vote, whether in person or by proxy or attorney, on a resolution at a meeting where that Member's vote is prohibited by the Act or an order of a court of competent jurisdiction.
- (e) The Organisation must disregard any vote on a resolution purported to be cast by a Member present at a meeting of Member where that person is not entitled to vote on that resolution, including by reason of Article 4.12(c) or (d).
- (f) The authority of a proxy or attorney for a Member to speak or vote at a meeting of Members is suspended while the Member who appointed the proxy or attorney is present in person at that meeting.

4.13 Objections to qualification to vote

- (a) An objection to the qualification of any person to vote at a meeting of Members may only be made:
 - (i) before that meeting, to the Board; or
 - (ii) at that meeting (or any resumed meeting if that meeting is adjourned), to the chair of that meeting.
- (b) Any objection under Article 4.13(a) must be decided by the Board or the chair of the meeting of Members (as the case may be), whose decision, made in good faith, is final and conclusive.

4.14 Proxies and powers of attorney

- (a) A Member, who is entitled to attend and cast a vote at a meeting of Members, may vote on a show of hands and on a poll:
 - (i) in person;
 - (ii) by not more than one proxy; or
 - (iii) by not more than one power of attorney.
- (b) A proxy or attorney of a Member must be a Member.
- (c) A Member may appoint a proxy or power of attorney for:
 - (i) all meetings of Members; or
 - (ii) any one or more specified meetings of Members.
- (d) An instrument appointing a proxy is valid if it is signed by the Member making the appointment and contains:
 - (i) the name and address of that Member;
 - (ii) the name of the Organisation;
 - the name of the proxy or the name of the office held by the proxy; and
 - (iv) the meeting(s) of Members at which the proxy may be used.
- (e) The chair of a meeting of Members may determine that an instrument appointing a proxy is valid even if it contains only some of the information specified in Article 4.14(d).
- (f) An instrument appointing a proxy or a power of attorney must be substantially in a form as may be prescribed in the By-laws or such other form as the chair of the meeting may accept from time to time.
- (g) Subject to the Act, the decision of the chair of a meeting of Members made in good faith as to the validity of an instrument appointing a proxy or power of attorney is final and conclusive.
- (h) Unless otherwise provided in the Act or in the appointment, a proxy or power of attorney may:
 - agree to a meeting of Members being called by shorter notice than is required by the Act or this Constitution;
 - agree to a resolution being either or both proposed and passed at a meeting of Members of which notice of less than the Prescribed Period is given;
 - (iii) speak on any resolution at a meeting of Members on which the proxy or power of attorney may vote;
 - (iv) vote at a meeting of Members (but only to the extent allowed by the appointment) on any resolution with respect to which the appointing Member may vote;
 - (v) demand or join in demanding a poll on any resolution at a meeting of Members on which the proxy or power of attorney may vote; and
 - (vi) attend and vote at any meeting of Members (at which the appointing Member may attend and vote) which is rescheduled or adjourned.
- (i) Unless otherwise provided in the Act or in the appointment, a proxy or power of attorney may vote on:
 - (i) any amendment to a resolution on which the proxy or power of attorney may vote;
 - (ii) any motion not to put that resolution or any similar motion; and
 - (iii) any procedural motion relating to that resolution, including a motion to elect the chair of a meeting of Members, vacate the chair or adjourn that meeting,

to the extent to which the appointing Member would be entitled to vote, even if the appointment does or does not direct the proxy or power of attorney how to vote on that resolution.

- (j) A proxy or an attorney for a Member cannot vote on a show of hands if they hold two or more appointments of proxy or power of attorney that specify different ways of voting.
- (k) With a notice of meeting of Members, the Organisation must send a form of proxy to Members in respect of that meeting of Members. The form of proxy is to be compliant with that prescribed in the By-laws and it must only provide for the Member:
 - to appoint a proxy of the Member's choice, but may specify who is to be appointed as proxy if the Member does not choose; and
 - (ii) to vote for or against each resolution, and may also provide for the Member to abstain from voting on one or more resolutions.
- (I) If the name of the proxy or the name of the office of the proxy in a proxy form of a Member is not filled in, the proxy of that Member is:
 - the person specified by the Organisation in the form of proxy in the case that the Member does not specify another person; or
 - (ii) if no person is so specified, the chair of that meeting.
- (m) A Member entitled to vote at a meeting of Members may specify the manner in which their proxy or attorney is to vote on a particular resolution at a meeting of Members but, unless specified, the proxy or attorney may vote as he or she thinks fit.
- (n) Subject to Article 4.14(j) and the provisions of this Article 4.14 generally, a proxy or power of attorney:
 - does not need to vote, unless the appointment specifies the way they must vote;
 - (ii) if the way they must vote is specified, they must vote in that way; and
 - (iii) if the proxy or attorney is also a Member, or holds proxy or attorney for more than one Member, may cast the votes held on a poll or ballot in different ways.
- (o) An appointment of proxy or power of attorney for a meeting of Members is effective only if the Organisation receives the appointment (and any authority under which the appointment was signed or a certified copy of the authority) not less than 48 hours before the time scheduled for commencement of that meeting (or any adjournment of that meeting).
- (p) Unless the Organisation has received notice in writing before the time scheduled for the commencement or resumption of a meeting of Members, a vote cast at that meeting by a person appointed by a Member as a proxy or power of attorney is, subject to this Constitution, valid even if, before the person votes, the appointing Member:
 - (i) dies; or
 - (ii) is mentally incapacitated; or
 - (iii) revokes the appointment of that person; or
 - (iv) revokes the authority under which the person was appointed by a third party.

5 Directors

5.1 Number and Qualification of Directors

- (a) The Organisation must have not less than nor more than:
 - (i) the number of Directors in aggregate specified in Item 2(a) of Schedule 2;
 - (ii) the number of Member Elected Directors specified in item 2(b) of Schedule 2; and
 - (iii) the number of Cricket Appointed Directors specified in item 2(d) of Schedule 2,

- and may have up to the number of Board Appointed Directors specified in item 2(c) of Schedule 2.
- (b) If the number of Directors or any class of Directors is below the number fixed by this Constitution the Board must promptly act and in a timely manner do all things reasonably within its power and ability, having regard to the provisions of this Constitution, to ensure the Organisation has the number of Directors for each class of Director required by this Article.
- (c) A Director must be a Member who is not disqualified from being a director of a corporation registered under the Act and who is entitled to unrestricted voting rights of a Member at an AGM.

5.2 Cricket Appointed Directors

- (a) Subject to Article 5,1(c), Cricket Appointed Directors may be appointed as follows:
 - 1 Cricket Appointed Director may be appointed by WADCC; and
 - (ii) 1 Cricket Appointed Director may be appointed by resolution of SGDC,

and of which notice of such appointment has been given to the Organisation under Article 13. A Cricket Appointed Director appointed in the terms of this Article may also be removed from their appointment as a Director by WADCC (in the case of a Cricket Appointed Director appointed under Article 5.2(a)(i)) or by SGDC (in the case of a Cricket Appointed Director appointed under Article 5.2(a) (ii)) notice of which removal has been notified to the Organisation under Article 13.

- (b) A Cricket Appointed Director will be appointed for a term commencing on the date of appointment in accordance with this Article 5.2 and ending on the earlier of the date of retirement or removal determined in accordance with Article 5.5(b) or 5.2(a).
- (c) A Cricket Appointed Director will be taken to have been appointed on the later of the date of receipt (or deemed receipt) by the Organisation of the notice of appointment mentioned in Article 5.2(a), and the effective date of appointment as specified in such a notice of appointment.
- (d) If a Cricket Appointed Director vacates or is removed from office for any reason, then the appointing body of that Cricket Appointed Director will promptly act to replace that Cricket Appointed Director in the terms of Article 5.2(a).

5.3 Appointment of Member Elected Directors by Members

- (a) Subject to Article 5.5(c), Member Elected Directors may be elected at general meetings of Members and are usually elected at the AGM of the Organisation.
- (b) A Member entitled to vote at a general meeting of Members may nominate a Member who is eligible under Article 5.1(c) to be a Member Elected Director.
- (c) The By-laws may prescribe the form and procedure to be followed with respect to:
 - (i) the nomination of a Member Elected Director;
 - (ii) the manner of presentation to the Members in general meeting of any such nomination for consideration by that meeting; and
 - (iii) the manner in which voting for Member Elected Directors will occur at the general meeting of Members, subject nevertheless to the Act.
- (d) Subject to Article 5.5(c) a Member Elected Director will be appointed for a term commencing on the date of the general meeting at which they were elected and ending on the date determined in accordance with Article 5.5(a), unless the Director is removed earlier in accordance with the Act or under Article 5.7(b).

5.4 Appointment of Board Appointed Directors

- (a) The Board may resolve to appoint Board Appointed Directors from time to time and also may resolve to remove a Board Appointed Director.
- (b) A Board Appointed Director will be appointed for a term commencing on the date of appointment in accordance with this Article 5.4 and ending on the earlier of the date of retirement or removal as determined in accordance with Article 5.5(b) or 5.4(a).
- (c) A Board Appointed Director appointed at a meeting of the Board will be taken to have been appointed on the date of that meeting or such later date as shall be the resolve of that meeting of the Board.
- (d) If a Board Appointed Director vacates or is removed from office for any reason, the Board may act to replace that Board Appointed Director in the terms of Article 5.4(a).

5.5 Retirement and re-appointment of Directors

- (a) Member Elected Director
 - Subject to this Article and to Schedule 3 (Transitional Provisions), a Member Elected Director must retire at the third AGM after the date of their election by the Members.
- (b) Board Appointed Director and Cricket Appointed Director. Subject to this Article and to Schedule 3 (Transitional Provisions), a Board Appointed Director and a Cricket Appointed Director must retire by at least the closure of the next meeting of the Board following the third anniversary of their appointment.
- (c) Casual Appointments (Member Elected Directors)
 - Subject to this Article, if a Member Elected Director vacates or is removed from office for any reason, then the Board may resolve to appoint another person as a Member Elected Director in the place of that Member Elected Director who is no longer continuing in office so as to ensure compliance with Article 5.1(b) PROVIDED THAT the term of that appointment will only be until the next AGM following the appointment, at which AGM the appointee must retire from office.
- (d) Subject to Article 5.6, a Director (including a Member Elected Director appointed under Article 5.5(c)) will be eligible for re-appointment (as applicable) at the expiration of their term of office.

5.6 Maximum term for a Director

- (a) Subject only to Articles 5.6(b) and Item 4 of Schedule 2, no Director may serve as a Director for more than the number of years in total specified in Item 3 of Schedule 2.
- (b) In the case of a person appointed to fill a casual vacancy on the Board as a Member Elected Director under Article 5.5(c) the period between the appointment of that person as a Director and the next AGM following that appointment, will not be counted as a Term or part of a Term and will not be included in determining whether a Director has served as a Director for the purposes of Article 5.6(a) or, in the case of the Chair for the purposes of Article 6.3(c).
- (c) For the purposes of this Article, any prior period of office served by a Director will not be counted towards the Director's maximum term of office if at least a continuous period of 3 years has elapsed during which the person was not a Director.

5.7 Vacation of office of Director

- (a) A Director may resign from office by giving the Organisation notice in writing.
- (b) Subject to the Act and Article 5.7(c), the Members in general meeting, by ordinary resolution, at any time may remove any Director from office.

- (c) A Director automatically ceases to be a Director if the Act so provides or if that Director:
 - (i) dies:
 - (ii) becomes of unsound mind or a person whose property is liable to be dealt with under a law relating to mental health;
 - (iii) is absent without the consent of the Board from all meetings of the Board held during a period of 3 months and the other Directors resolve that his or her office be vacated:
 - (iv) resigns or is removed from office under this Constitution;
 - (v) becomes ineligible to be a Director either under this Constitution, the By-laws or at law;
 - (vi) becomes an undischarged bankrupt;
 - (vii) ceases to be a Member.

5.8 No Alternate Directors

A Director may not appoint a person as his or her alternate Director

5.9 Payment to Directors prohibited

- (a) Subject to Article 5.9(b), the Organisation must not pay any fees to a Director for performing his or her duties and responsibilities as a Director unless the Members in general meeting have approved the payment or the Members in general meeting have approved payment of fees to Directors, at the discretion of the Board, with a maximum annual pool or cap amount for all Directors in any one year, and the payment is within that annual pool or cap amount.
- (b) A payment of the kind referred to in Article 1.5(b) may be made to a Director if that payment has been approved by the Board.

5.10 Skills, experience and attributes of Directors

(a) General skills, experience and attributes

When Members are nominating Member Elected Directors, when the Members are electing Member Elected Directors, when the Board is making casual appointees under Article 5.5(c) or appointing Board Appointed Directors and when WADCC or SGDC are appointing Cricket Nominated Directors, they are encouraged to have regard to the skills, experience and attributes of the Board as a whole, and those skills, experience and attributes that are best likely to lead the Organisation in the pursuit of its strategies and objects and to assure the proper governance oversight of management. Without limitation or prescription, those skills, experience and attributes of the Board as a whole to which Members and the Board should have regard include:

- relevant business, professional, financial and governance experience, skills and acumen;
- an understanding of and empathy with relevant stakeholder interests of the Organisation including without limitation being currently active and engaged (whether in a playing, administration or other capacity) within country, premier and/or community cricket;
- (iii) an understanding of the game and business of cricket:
- (iv) experience in the governance and management of sporting or community based organisations; and
- (v) diversity of experience and perspective.
- (b) Board Skills Matrix (or similar)

From time to time the Board (including via a Board committee with delegated authority for this purpose) should determine, in its judgement, and communicate to the Members, at or around the time when nominations for Member Elected Directors are open:

- the desired suite of skills, experience and attributes that the Board (as a whole) desirably should have to meet the Organisation's strategic objectives and needs (including by way of guidance only, to those listed in (a) above);
- (ii) any "gap" in the desired skills, experience and attributes of the existing and continuing Directors so as to assist Members in identifying prospective nominees, and to assist Members in the exercise of their voting for Member Elected Directors.
- (c) Background Information on proposed Member Elected Directors, Board Appointed Directors and Cricket Appointed Directors.
 - (i) Any proposed nominee for the election or appointment as a Director must provide background career and personal information as to their skills, experience and attributes to the Board (including via a Board committee with delegated authority for that purpose) as provided for in the By-laws.
 - (ii) With respect to prospective Member Elected Directors, this information, or following consultation with the nominee, a succinct summary of it, is to be provided by the Board (or its committee) to the Members, including at or about the time of election of Member Elected Directors so as to assist Members in the exercise of their voting for Member Elected Directors.

5.11 Consents and Undertakings

Each Director must sign, complete and deliver to the Organisation such forms of consent, contact information and undertakings to observe and comply with the Act, the Constitution and the By-laws, as may be prescribed in the By-laws, as a precondition to their taking office as a Director.

6 Officers and Officeholders

6.1 Patron-in-Chief

- (a) On recommendation of a resolution of the Board and by resolution of the Members in general meeting, the Organisation may from time to time invite a suitably credentialed dignitary to accept the position of "Patronin-Chief" of the Organisation.
- (b) Upon the issue of such an invitation and its acceptance, the dignitary shall become the "Patron-in-Chief" of the Organisation.
- (c) The Patron-in-Chief shall hold office until the earlier of:
 - the Patron-in-Chief giving written notice to the Organisation resigning from the office;
 - (ii) the Board resolving by Absolute Special Majority to remove the Patron-in-Chief from office.
- (d) The role and responsibilities of the Patron-in-Chief will be of a ceremonial nature as may be prescribed in the By-laws.

6.2 Patron(s)

- (a) On recommendation of a resolution of the Board and by resolution of the Members in general meeting, the Organisation may from time to time invite up to 3 eminent leaders of the game of cricket in Western Australia to accept a position as a "Patron" of the Organisation.
- (b) Upon issue of such invitation and its acceptance, the relevant person shall become a "Patron" of the Organisation.
- (c) A Patron shall hold office until the earlier of:
 - the next AGM of the Organisation following the 5th anniversary of their appointment as a Patron;

- (ii) a Patron giving written notice to the Organisation resigning from the office of a Patron; and
- (iii) the Board resolving by Absolute Special Majority to remove that Patron from office.
- (d) A Patron retiring from office under Article 6.2(c)(i) is eligible for re-appointment to office as a Patron.
- (e) The role and responsibilities of a Patron will be of a ceremonial and promotional nature as may be prescribed in the By-laws.

6.3 Chair and Deputy Chair

- (a) Subject to this Article 6.3, the Board shall:
 - elect a Director as Chair of the Board for a term of three years (or such shorter period determined by the Board), or until the relevant person ceases to be a Director; and
 - (ii) elect a Director as Deputy Chair, for a term of three years (or such shorter period determined by the Board) or until the relevant person ceases to be Director.
- (b) The Board may remove:
 - (i) the Chair as chair of the Board, and/or
 - (ii) the Deputy Chair as deputy chair of the Board,
 - by Board resolution at any time.
- (c) A Director must not be elected as the Chair for more than the number of years in total specified in Item 4(a) of Schedule 2 and must not be elected as the Chair if the term of appointment as Chair would result in the Director serving on the Board (whether as a Director or the Chair) for more than the number of years in total specified in Item 4(b) of Schedule 2 (or such longer period where Article 5.6(b) applies).
- (d) The Chair must (if present within 15 minutes after the time appointed for the holding of the meeting and willing to act) chair each meeting of the Board.
- (e) If:
 - (i) there is no Chair; or
 - the Chair is not present within 15 minutes after the time appointed for the holding of a meeting of the Board; or
 - the Chair is present within that time but is not willing or able to chair all or part of that meeting,

then the Deputy Chair, or in default of the Deputy Chair, one of the Directors present elected by those present, will chair the meeting.

6.4 Chief Executive Officer

- (a) The Board may appoint a person as the CEO.
- (b) Subject to any agreement between the Organisation and the CEO, the Board may remove or dismiss or suspend the CEO at any time, with or without cause.
- (c) The Directors may delegate such of their powers (including the power to delegate) to the CEO as provided in Article 7.3.
- (d) Subject to any agreement between the Organisation and the CEO the Board may:
 - (i) revoke or vary the appointment of the CEO;
 - (ii) revoke or vary any power delegated to the CEO; and/or
 - (iii) suspend the appointment of the CEO or suspend the delegation of powers to the CEO for such period and on such terms as the Board may determine in good faith.
- (e) The CEO must exercise the powers delegated to him or her in accordance with any lawful directions of the Board.

- (f) The exercise of a delegated power by the CEO is as effective as if the Board exercised the power.
- (g) The CEO is not a Director and has no inherent power or right to attend, speak or vote at meetings of the Board, however by standing or other invitation of the Chair, the CEO may and is expected to attend and speak at meetings of the Board, other than when the Board may be meeting in "closed session" without executives and/or management present.
- (h) The CEO is not entitled to hold office as a Director.

6.5 Secretary(s)

- (a) The Organisation must have at least one Secretary as required by the Act.
- (b) A Secretary or up to 2 Secretaries for the Organisation may be appointed by the Board for any period and on any terms (including as to remuneration) as the Board resolve.
- (c) Subject to any agreement between the Organisation and a Secretary, the Board may remove or dismiss a Secretary at any time, with or without cause, and may revoke or vary the appointment of a Secretary.
- (d) The Board will determine the role, duties and responsibilities of a Secretary and to whom a Secretary will account, including as may be provided for in the By-laws.

6.6 Indemnity and insurance

- (a) To the extent permitted by law, the Organisation must indemnify each Relevant Officer against:
 - (i) a Liability of that person; and
 - (ii) Legal Costs of that person.
- (b) To the extent permitted by the law, the Organisation may make a payment (whether by way of advance, loan or otherwise) to a Relevant Officer in respect of Legal Costs of that person.
- (c) To the extent permitted by the law, the Organisation may pay, or agree to pay, a premium for a contract insuring a Relevant Officer against:
 - (i) a Liability of that person; and
 - (ii) Legal Costs of that person.
- (d) To the extent permitted by law, the Organisation may enter into an agreement or deed with:
 - (i) a Relevant Officer; or
 - (ii) a person who is, or has been, an officer of the Organisation or a related body corporate of the Organisation,

under which the Organisation must do all or any of the following:

- (iii) keep books of the Organisation and allow that officer, and his or her advisers, access to those books on the terms agreed;
- (iv) indemnify that officer against any Liability of that officer:
- (v) make a payment (whether by way of advance, loan or otherwise) to that officer in respect of Legal Costs of that officer; and
- (vi) keep that officer insured in respect of any act or omission by that officer, while a Relevant Officer or an officer of the Organisation or a related body corporate of the Organisation, on the terms agreed (including as to payment of all or part of the premium for a contract of insurance).

7 Powers of the Organisation and Board

7.1 General powers

- (a) Subject to this Constitution, the Organisation may exercise, in any manner permitted by the Act, any power which a public company limited by guarantee may exercise under the Act in order to achieve the objects and strategic focus of the Organisation.
- (b) The Board is responsible for the sound governance of the Organisation.
- (c) Subject to this Constitution, the business of the Organisation is to be managed by, or under the direction of, the Board, and the Board may exercise all the powers of the Organisation except any powers that the Act or this Constitution requires the Organisation to exercise in General Meeting or only with the approval of Members in General Meeting.
- (d) Other than with the approval of Members in General Meeting, the Board shall not:
 - (i) sell or dispose of any land of the Organisation;
 - (ii) purchase or acquire any land for the Organisation;
 - (iii) mortgage, charge, pledge or encumber by way of security by any means whatsoever or howsoever any land or other assets or income of the Organisation; or
 - (iv) "deal with the WACA Ground", within the meaning and context of Article 1.4(a)(iii) of this Constitution, if to do so would prejudice the WACA Ground as a premier venue for hosting games of cricket;

PROVIDED FURTHER THAT should the Board propose to do any matter or thing requiring the approval of Members in General Meeting in the terms of this Article 7.1(d) then the Board must provide to the Members sufficient relevant information concerning the proposed matter to enable the Members to make a reasonably informed decision as to the proposed matter.

7.2 Execution of Documents

- (a) If the Organisation has a common seal:
 - the Organisation may execute a document if that seal is fixed to the document and the fixing of that seal is countersigned by:
 - (A) two Directors; or
 - (B) a Director and a Secretary; or
 - (C) a Director and another person appointed by the Board for that purpose.
 - (ii) particulars of the use of the common seal (including the reasons for its use, its counter-signatories and the date of its use) will be promptly recorded in the Organisation's corporate secretarial records.
- (b) The Organisation may execute a document without a common seal if the document is signed by:
 - (i) two Directors; or
 - (ii) a Director and a Secretary; or
 - (iii) a Director and another person appointed by the Board for that purpose.
- (c) The Board may resolve, generally or in a particular case:
 - that any signature on certificates or cards for Membership, or other common use documents approved of by the Board, may be affixed by mechanical, electronic or other means;
 - to delegate the power to execute documents on behalf of the Organisation to officers of the Organisation;

- (iii) to grant and execute power(s) of attorney to execute documents and to do such things on behalf of and in the name of the Organisation to such person or persons, as the Board may determine.
- (d) Negotiable instruments may be signed, drawn, accepted, endorsed or otherwise executed by or on behalf of the Organisation in the manner, and by the persons, as the Board determines resolves.

7.3 Committees and delegates

- (a) The Board may delegate any of its powers (including this power to delegate) to a committee of the Board, the CEO, an employee of the Organisation or a Member.
- (b) The Board may revoke or vary any power delegated under Article 7.3(a).
- (c) A committee or delegate must exercise the powers delegated in accordance with any directions of the Board.
- (d) The proper exercise of a delegated power by the committee or delegate is as effective as if the Board exercised the power itself.
- (e) Any delegation of the Board of any of its powers must be recorded in the Board's minute book.

7.4 Attorney or agent

- (a) The Board may appoint any person to be attorney or agent of the Organisation for any purpose, for any period and on any terms (including as to remuneration) as the Board resolves.
- (b) The Board may delegate any of their powers (including the power to delegate) to an attorney or agent.
- (c) The Board may revoke or vary:
 - (i) an appointment under Article 7.4(a); or
 - (ii) any power delegated to an attorney or agent.

7.5 Affiliated Organisations

- (a) The Board may by Absolute Special Majority resolve to declare organisations relevant to the game of cricket to be afforded the status of Affiliated Organisations.
- (b) The Members by resolution in General Meeting, and on the recommendation of the Board by resolution passed by an Absolute Special Majority, may determine that an Affiliated Organisation no longer holds status as an Affiliated Organisation.
- (c) An Affiliated Organisation may at any time by notice in writing to the Organisation declare that it no longer holds status as an Affiliated Organisation.
- (d) The Organisation may enter into agreements from time to time with Affiliated Organisations including, and without limitation, with respect to support concerning:
 - (i) funding;
 - (ii) logistics, support;
 - (iii) training;
 - (iv) equipment and facilities;
 - (v) services;
 - (vi) co-operative arrangements;
 - (vii) governance and policies;
 - (viii) otherwise.
- (e) Any termination of status of an Affiliated Organisation under this article is without prejudice to any contractually accrued or accruing rights under an agreement entered into in the terms of (d) above.

7.6 Duties of Directors

The Directors must comply with their duties as directors under the law, including the duties imposed on directors and officers under the Act.

7.7 Conflicts of Interest

- (a) A Director must disclose the nature and extent of any actual or perceived material conflict or interest in a matter that is being considered at a meeting of the Board (or that is proposed in a circular resolution):
 - (i) to the other Directors;
 - (ii) if all of the Directors have the same conflict of interest, to the Members at the next meeting of Members, or at an earlier time if reasonable to do so.
- (b) The disclosure of a conflict of interest by a Director must be recorded in the minutes of the meeting.
- (c) Each Director who has a material personal interest in a matter that is being considered at a meeting of the Board (or that is proposed in a circular resolution) must not, except as provided under Article 7.7(d):
 - be present at the meeting while the matter is being deliberated; or
 - (ii) vote on the matter.
- (d) A Director may still be present and vote if:
 - their interest arises because they are a Member of the Organisation, and the other Members have the same interest:
 - their interest relates to an insurance contract that insures, or would insure, the Director against liabilities that the Director incurs as a Director of the Organisation;
 - (iii) their interest relates to a payment by the Organisation under an indemnity provided in accordance with Article 6.6 or any contract relating to an indemnity that is allowed under the Act;
 - (iv) the Australian Securities and Investments Commission makes an order allowing the Director to vote on the matter: or
 - (v) the Directors who do not have a material personal interest in the matter pass a resolution that:
 - identifies the Director, the nature and extent of the Director's interest in the matter and how it relates to the affairs of the Organisation; and
 - (B) says that those Directors are satisfied that the interest should not stop the Director from voting or being present.

7.8 Cricket Australia

The Board and the Organisation must do all things reasonable necessary and desirable to seek to achieve the objects listed in Article 1.4(a)(ii).

8 Proceedings of Board

8.1 Written resolutions of the Directors

- (a) The Directors may pass a resolution, without a meeting of the Board being held, if a copy of the proposed resolution is sent to all Directors and not less than an Absolute Special Majority of Directors, entitled to vote on the resolution, assent to a document containing a statement that they are in favour of the resolution set out in the document.
- (b) Separate copies of documents referred to in Article 8.1(a) may be used for assenting to by Directors if the wording of the resolution and the statement is the same in each copy.
- (c) A Director may signify assent to a document under Article 8.1(a) by signing the document or by notifying the Organisation of that assent in a manner permitted by Article 13.3.

- (d) Where a Director signifies assent to a document under Article 8.1(c) other than by signing the document, the Director must, by way of confirmation, sign the document before, or at, the next Board meeting attended by that Director.
- (e) The resolution the subject of a document under this Article is not invalid if a Director does not comply with Article 8.1(d).

8.2 Meetings of the Board

- (a) The Directors may meet, adjourn and otherwise regulate their Board meetings as they think fit.
- (b) A meeting of the Board may be held, including remotely at more than one venue, using any technology consented to by all Directors.
- (c) The consent of the Directors under Article 8.2(b) may be for all meetings of the Board or for any one or more specified meetings and a Director may within a reasonable period before a meeting by written notice to the Organisation, withdraw a consent previously given.
- (d) If a meeting of the Directors is held in two or more places linked together by any technology:
 - a Director present at one of the places is taken to be present at the meeting unless and until that Director states to the chair of the meeting that he or she is discontinuing participation in the meeting; and
 - (ii) the chair of that meeting may determine at which place the meeting will be taken to have been held.

8.3 Who can call meetings of the Board

- (a) The Chair, the Deputy Chair or any 2 other Directors jointly may call a meeting of the Board at any time.
- (b) On a request under Article 8.3(a), a Secretary (or another acting in the role of the Secretary) must call a meeting of the Board.

8.4 How to call meetings of the Board

- (a) Subject to Article 8.4(e), notice of a meeting of the Board must be given to each Director.
- (b) A notice of meeting of the Directors must:
 - (i) set out the place, date and time for the meeting (and, if the meeting is to be held in two or more places, the technology that will be used to facilitate this); and
 - (ii) state the general nature of the business of the meeting. $\label{eq:continuous} % \begin{subarray}{ll} \end{subarray} % \begin{subarray}{ll$
- (c) Subject to Article 8.4(e), not less than 24 hours' notice of a meeting of the Board must be given, unless an Absolute Special Majority of Directors agree otherwise.
- (d) A Director may waive notice of a meeting of the Board by notice in writing to the Organisation to that effect.
- (e) In the case of an emergency or other corporate urgency, a meeting may be called on short notice, without formal notice of meeting, at the behest of the Chair, the Deputy Chair or the CEO.

8.5 Quorum

- (a) Subject to the Act, a quorum for a meeting of the Board is a majority of the total number of Directors.
- (b) A quorum for a meeting of the Board must be present at all times during the meeting PROVIDED THAT a person who has declared a conflict of interest and has absented themselves (or abstained from deliberating and voting) on a matter before the meeting will still be counted as being present for the purposes of a quorum.

8.6 Resolutions of the Board

- (a) A resolution of the Board, other than a resolution requiring an Absolute Special Majority, is passed if more votes are cast in favour of the resolution than against it.
- (b) Subject to the Act and this Article 8.6, each Director has one vote on a matter arising at a meeting of the Board.
- (c) Subject to the Act, in case of an equality of votes on a resolution at a meeting of the Board, the chair of that meeting has a casting vote on that resolution in addition to any vote the chair of the meeting has in his or her capacity as a Director, except where there are only 2 Directors entitled to vote on the resolution.

9 Board and other Committees

9.1 Board may establish Committees

The Board may on its own resolution constitute committees of the Board to assist it in the performance of its responsibilities.

9.2 Terms of Reference and Governance

The terms of reference, governance arrangements, and meeting processes and accountability requirements of Board committees will be determined by the By-laws or by the Board at the time of establishment of the relevant committee.

9.3 Review

The Board may from time to time:

- (a) review and vary the matters referred to in Article 9.2; and
- (b) resolve to terminate the ongoing constitution of a Board committee.

9.4 SGDC

There shall be constituted by virtue of this Article and the terms of the Governance Charter incorporated as part of the By-laws (refer By-law 2.1 in Schedule 4), a committee to be known as the Statewide Game Development Committee whose terms of reference and membership shall be prescribed under the By-laws PROVIDED THAT membership of the Committee shall include an appointee from each Affiliated Organisation and other cricket related advisory committees of the organisation which are endorsed by the Board and which represent the interests of discrete sections of the cricketing community, with such appointees being the only voting members of the Committee, with power and authority for the Committee in the terms of Article 5.2 concerning Cricket Appointed Directors.

10 By-laws

10.1 By-laws

In addition to the provisions of this Constitution including Article 10.6 and Schedule 4, and subject to Article 10.3, By-laws may from time to time be prescribed to govern and/or regulate By-law Matters.

10.2 By-law Matters

By-law matters include:

- (a) any matter of a nature authorised by this Constitution to be capable of being the subject of a By-law;
- (b) any matter pertinent to:
 - (i) Membership:
 - (ii) Fees;
 - (iii) meetings of Members:
 - (iv) rights, responsibilities and conduct of Members;
 - (v) procedures concerning nomination, voting on and election of Member Elected Directors
 - (vi) governance and management of the Organisation;
 - (vii) policies, procedures, protocols and practices of the Organisation.

10.3 No conflict with Constitution

No By-law may conflict with a provision of this Constitution and if it does:

- (a) the By-law is either void or will be read down to the extent of such conflict; and
- (b) the provisions of this Constitution will prevail.

10.4 Making of By-laws

- (a) Other than as provided in Article 10.6, By-laws may be made, varied or revoked by resolution of the Board passed with not less than an Absolute Special Majority of the Board.
- (b) Other than as provided in Article 10.6, By-Laws, and any variation to or revocation of them, will become effective on the date so determined by the Board at the time of their making or in the absence of such a determination, 30 days after the Board resolution.
- (c) Within 7 days of the Board making, varying or revoking a By-Law, a copy of the By-Law (or its variation or revocation, as applicable) must be published on the Organisation's website in a reasonably prominent location to inform Members thereof with notice being given to all Members of the change that has been made.

10.5 Force and Effect of By-Laws

By-Laws will have contractual force and effect between Members inter se, between the Organisation and the Members, and between the Organisation and its officers, in the same manner as if the By-laws had constitutional force and effect in their own right.

10.6 Current By-Laws upon adoption of Constitution

The By-laws to apply from the Effective Date of adoption of this Constitution are set out or described in Schedule 4.

11 Records

11.1 Financial Year

The financial year of the Organisation ends on 30 June in each year.

11.2 Accounts

The Board, through the Secretary and the CEO, shall cause proper accounts, books and records to be kept, in accordance with accepted Australian accounting standards consistently applied for like bodes corporate to the Organisation and otherwise as may be required under the Act with respect to:

- (a) all sums of money received or expended by the Organisation and the matter in respect of which the receipt or expenditure takes place;
- (b) all financial transactions of the Organisation;
- (c) the assets and liabilities of the Organisation; and
- (d) the Organisation's operations.

11.3 Books, accounts and records

The accounts, books and records shall be kept at the office of the Organisation or such other place from time to time determined by the Board and are open to inspection by the Directors and must be retained in accordance with the Act.

11.4 Audit

The Board shall cause the accounts to be audited annually by the Auditor.

11.5 Minutes

- (a) The Organisation must, within one month, make and keep the following records:
 - minutes of proceedings and resolutions of meetings of Members or Directors' meetings (including meetings of any committees);
 - (ii) minutes of circular resolutions of Directors;

- (iii) a copy of a notice of each meeting of Members of Directors' meetings; and
- (iv) a copy of any members' statement distributed to members in accordance with Article 4.1(d).
- (b) The Directors must ensure that minutes of a meeting of Members or a Directors' meeting are signed within a reasonable time after the meeting by:
 - (i) the chair of the meeting; or
 - (ii) the chair of the next meeting.
- (c) Records of minutes of meetings of Members are open for inspection by Members.

12 Alterations to Constitution

No new Article or provision shall be added to this Constitution nor shall any of the Articles or provisions contained herein be amended, altered or rescinded unless:

- (a) a Special Resolution of Members in general meeting so to do is passed: and
- (b) the provisions of the Act and this Constitution are complied with.

13 Notices

13.1 Notice to Members

- (a) Subject to Article 13.1(b), the Organisation may give notice to a Member:
 - (i) by hand delivery;
 - (ii) by sending it by prepaid post to the address of the Member in the Register or the alternative address (if any) nominated by that Member:
 - (iii) by sending it to the fax number or electronic or digital address (if any) nominated by that Member; or
 - (iv) with an Absolute Special Majority of the Board, by advertisement in accordance with Article 13.1(c).
- (b) If the address of any Member in the Register is not within Australia (and its Territories) and that Member does not nominate an alternative address within Australia (and its Territories), unless otherwise specified by the Act, the Organisation may (in addition to any method of service specified in Article 13.1(a)) give a notice to that Member by:
 - posting it on the Organisation's internet website (if any); or
 - (ii) advertisement in accordance with Article 13.1(c).
- (c) Any notice allowed to be given by the Organisation to Members by advertisement is sufficiently advertised if advertised:
 - (i) in a newspaper published on most days and generally circulating in Western Australia: or
 - (ii) in electronic form on an online media portal reasonably approved of by the Board and which is generally available and accessible to members of the public in Western Australia (even if a subscription fee is payable).

13.2 Notice to Directors

The Organisation may give notice to a Director:

- (a) by hand delivery;
- (b) by sending it by prepaid post to the usual residential address of that person or the alternative address (if any), including a post office box address, nominated by that person;
- (c) by sending it to the fax number or electronic or digital address (if any) nominated by that person; or
- (d) by any other means agreed between the Organisation and that person.

13.3 Notice to the Organisation

A person may give notice to the Organisation:

- (a) by leaving it at the registered office of the Organisation;
- (b) by sending it by prepaid post to the registered office of the Organisation;
- (c) by sending it to the fax number at the registered office of the Organisation:
- (d) by sending it to the electronic or digital address (if any) nominated by the Organisation for that purpose; or
- (e) by any other means permitted by the Act.

13.4 Time of service

- (a) A notice sent by prepaid post to an address within Australia is taken to be given:
 - in the case of a notice of meeting, two Business Days after it is posted; or
 - in any other case, at the time at which the notice would be delivered in the ordinary course of post.
- (b) A notice sent by prepaid post to an address outside Australia is taken to be given:
 - (i) in the case of a notice of meeting, four Business Days after it is posted; or
 - (ii) in any other case, at the time at which the notice would be delivered in the ordinary course of post.
- (c) A notice sent by fax or electronic or digital means is taken to be given on the Business Day it is sent, provided that the sender's transmission report shows that the whole notice was sent to the correct fax number or electronic or digital address.
- (d) A notice posted on the Organisation's internet website or advertised in accordance with Article 13.1(c) is taken to be given on the date the notice is first published on the Organisation's website or the date on which it is first advertised in terms of Article 13.1(c) (as applicable).
- (e) The giving of a notice by prepaid post is sufficiently proved by evidence that the postage was paid and the notice:
 - (i) was addressed to the correct postal address of the recipient; and
 - (ii) was placed in the post.

13.5 Signatures

The Board may decide, generally or in a particular case, that a notice given by the Organisation be signed by mechanical, electronic or other means.

14 Winding up

14.1 Transfer of surplus

Subject to the Act and any other applicable laws or court orders, on a winding up of the Organisation, the Members must determine one or more companies, associations or institutions whose constitutions:

- (a) require them to pursue only objects similar to those in Article 1.4 and to apply their income in promoting those objects; and
- (b) prohibit them from making payments or distributions to their members to at least the same extent as in Article 1.5, to whom the liquidator must give or transfer any surplus assets or property on winding up in accordance with a distribution plan also determined by Members.

14.2 Application to Court

If the Members fail to make a determination under Article 14.1 within 20 Business Days of the winding up of the Organisation, the liquidator must make an application to the Court to make that determination.

15 Transitional Provisions

15.1 Transitional Arrangements

The Transitional Provisions shall apply in accordance with their terms as if set out in this Constitution in full.

15.2 Transitional Provisions Paramount

In the case of any inconsistency between the Transitional Provisions and any other provision of this Constitution or the By-laws, the Transitional Provisions will prevail.

15.3 Continuance of Transitional Provisions

The Transitional Provisions shall have continuing force and effect in accordance with their terms.

Definitions and interpretation

1. Definitions

In this Constitution:

"Absolute Special Majority" in the context of a Board resolution means a resolution passed by not less than 75% of the total number of Directors (excluding Directors who have declared a conflict of interest concerning the matter the subject of the proposed resolution and who as a consequence do not vote on the matter) holding office at the time of the resolution.

"Act" means the Corporations Act 2001 (Commonwealth).

"AGM" means the Annual General Meeting of Members.

"Affiliated Organisation" means an organisation (incorporated or unincorporated) involved in or associated with the administration, co-ordination and/or playing of cricket and which is determined by to be an Affiliated Organisation in the terms of Article 7.5 or is listed in Item 5 of Schedule 3 (Transitional Provisions) and which has not been determined to no longer have status as an Affiliated Organisation.

"Applicable Form" has the meaning ascribed to it in the By-laws.

"Board" means the Directors meeting from time to time as a board of directors of the Organisation.

"Board Appointed Director" means a Director appointed pursuant to Article 5.4.

"Business Day" means a day except a Saturday, Sunday or public holiday in Western Australia.

"By-Laws" means by-laws of the Organisation in the terms of Article 10. $\,$

"By-Law Matters" means matters of the nature mentioned in Article 10.2 which may be the subject of a By-Law.

"Cessation Event" means, in respect of a Member:

(a) the death or bankruptcy of that Member; or

(b) that Member becoming of unsound mind or a person whose property is liable to be dealt with under a law relating to mental health.

"Chair" means a person elected as chair of the Board under Article 6.3.

"CEO" means the person appointed for the time being under Article 6.4, being an employee (whether full-time or part-time) of the Organisation or a related body corporate of the Organisation.

"Class" means a class of Membership as prescribed in the By-Laws.

"Cricket Appointed Director" means a Director appointed pursuant to Article 5.2:

"Cricket Australia" means the peak organisation for the promotion, development, control, co ordination and management of the sport of cricket in Australia, with Cricket Australia ABN 53 006 089 130 being acknowledged as that organisation:

"Director" means a director of the Organisation for the time being.

"Effective Date" has the meaning given in Schedule 3.

"Expulsion Event" means, in respect of a Member:

(a) that Member has been convicted within the preceding 5 years in criminal proceedings brought in connection with a contravention of the law where the penalty for the offence may be incarceration for a period in excess of 12 months, but excluding any "spent conviction";

- (b) that Member has failed to pay any Fees or other money owing to the Organisation and those Fees or other moneys remain outstanding for at least 40 Business Days after written notice of demand for payment has been issued to the Member; or
- (c) the Board determines that in its opinion exercised in good faith:
 - the conduct of that Member is unbecoming of a Member; or
 - (ii) that the continued Membership of the Member may be prejudicial to the objects, interests or reputation of the Organisation: or
 - (iii) the Member is in material or persistent breach of the By-laws or this Constitution.

"Fees" means a fee or levy payable by Members under Article 3.1.

"Financial Year" means 1 July in one year to 30 June (inclusive) in the next succeeding year, or such other period as may be prescribed in the By-Laws.

"General Meeting" means a general meeting of Members duly called and held in the terms of Article 4.1 and includes an AGM and an extraordinary general meeting;

"Legal Costs" for a person, means legal costs incurred by that person in defending an action for a Liability of that person.

"Liability" of a person, means a liability incurred by that person as an officer of the Organisation or a related body corporate of the Organisation.

"Member" means a natural person whose name is entered in the Register as a member of the Organisation and "Membership" has a corresponding meaning.

"Membership Card" means a card issued to a Member under Article 2.8.

"Member Elected Director" means a Director appointed pursuant to Article 5.3.

"Organisation" has the meaning in Article 1.2(a).

"Prescribed Notice" means the Prescribed Period of notice or any shorter period of notice for a meeting allowed under the Act.

"Prescribed Period" means 21 days.

 $\begin{tabular}{ll} \begin{tabular}{ll} \beg$

"Relevant Officer" means a person who is, or has been, a Secretary, a Chief Executive Officer, a Director or another person who has held office in the Organisation and would be defined as an "officer" in the terms of the Act.

"Secretary" "Secretary(s)" and **"Secretaries"** means the company secretary and/or secretaries of the Organisation for the time being.

"Special Resolution" has the meaning ascribed to it in the Act. "SGDC" means the standing committee of the Organisation constituted in the terms of Article 9.4.

"Term" means the term of office of each Director in accordance with these Articles.

"Transitional Provisions" means those provisions of this Constitution appearing in Schedule 3.

"WACA Ground" means the sporting complex and associated buildings at Nelson Crescent, East Perth, Western Australia being Lot 1507 on Deposited Plan 410829 and being the whole of the land in Certificate of Title Volume 2938 Folio 506.

"WADCC" means WA District Cricket Council Inc, as an Affiliated Organisation.

2. Interpretation

- (a) In this Constitution:
 - a reference to a meeting of Members includes a meeting of any class of Members;
 - a Member is taken to be present at a meeting of Members if the Member is present in person or by proxy, attorney or body corporate representative; and
 - (iii) a reference to a notice or document in writing includes a notice or document given by fax or other form of written communication.
- (b) In this Constitution, headings are for convenience only and do not affect interpretation.
- (c) Unless the context indicates a contrary intention, in this Constitution:
 - words importing the singular include the plural (and vice versa);
 - (ii) words indicating a gender include any other gender and any non specific gender affiliations;
 - (iii) the word "person" includes an individual, the estate of an individual, a corporation, an authority, an association or a joint venture (whether incorporated or unincorporated), a partnership and a trust;
 - (iv) where a word or phrase is given a defined meaning, any other part of speech or grammatical form of that word or phrase has a corresponding meaning; and
 - (v) the word "includes" in any form is not a word of limitation.
 - (vi) a reference to an Article or a Schedule, is to an article or a schedule of this Constitution;
 - (vii) a reference in a Schedule to a paragraph is to a paragraph of that Schedule;
 - (viii) a Schedule is part of this Constitution;
 - (ix) a reference to this Constitution, is to this Constitution (and where applicable any of its provisions) as modified or repealed from time to time; and
 - (x) if there is more than one Secretary appointed, the receipt or giving of a notice to or by a Secretary, or the signing of a notice or document by a Secretary, or the exercise of any power under this Constitution by or in the name of the Secretary, is effective and binding if done by a Secretary;
- (d) Unless the context indicates a contrary intention, in this Constitution, a reference to any statute or to any statutory provision includes any statutory modification or re-enactment of it or any statutory provision substituted for it, and all ordinances, by-laws, regulations, rules and statutory instruments (however described) issued under it.
- (e) Unless the context indicates a contrary intention, in this Constitution:
 - an expression that deals with a matter dealt with by a provision of the Act has the same meaning as in that provision:
 - (ii) an expression that is defined in Section 9 of the Act has the same meaning as in that Section.

3. Exercise of powers

Where this Constitution confers a power or imposes a duty, then, unless the contrary intention appears, the power may be exercised and the duty must be performed, from time to time, as the occasion requires.

4. Severing invalid provisions

If at any time any provision of this Constitution is or becomes illegal, invalid or unenforceable in any respect under the law of any jurisdiction, that does not affect or impair:

- (a) the legality, validity or enforceability in that jurisdiction of any other provision of this Constitution; or
- (b) the legality, validity or enforceability under the law of any other jurisdiction of that or any other provision of this Constitution

Key Information

Quorum of Members for Meetings (refer Article 4.6)

(a) General meetings of Members

The lesser of 40 Members with unrestricted voting rights and 5% of the total number of Members with unrestricted voting rights at the time the relevant meeting is held.

(b) Class meeting of designated Membership Classes

The lesser of 20 Class Members with unrestricted Class voting rights, and 5% of the total number of Members in that Class with unrestricted voting rights at the time the relevant meeting is held.

2. Number of Directors (refer Article 5)

(a) Number of Directors

Seven (7) plus the number of Board Appointed Directors

(b) Number of Member Elected Directors

Five (5)

(c) Number of Board Appointed Directors

Three (3)

(d) Number of Cricket Appointed Directors

Two (2)

3. Maximum term as a Director (refer Article 5.6)

9 years

Maximum periods for service as Chair (refer Article 6.3(c))

- (a) Maximum term as Chair 6 years(*)
- (b) Maximum total combined years of appointment as a Director and as the Chair 12 years(*)
- (*) PROVIDED THAT on recommendation of the Board (passed by an Absolute Special Majority) and by resolution of Members in General Meeting, these maxima may be extended for a further period of 12 months in circumstances where it is considered to be in the best interests of the Organisation and its succession planning to do so.

Transitional Provisions

- 1. For the purposes of these Transitional Provisions:
 - (a) "Effective Date" means the date of registration of the Organisation under the Act.
- Terms uniquely defined in the constitution of Western Australia Cricket Association (Inc) and used in these Transitional Provisions will have those defined meanings. Subject to paragraph 1(b) preceding, from the Effective Date the existing constitution, by-laws and standing orders that previously applied to the Organisation are of no further force or effect.
- With respect to those holding constitutionally recognised office as at the Effective Date, the following provisions shall apply:
 - (a) Subject to paragraph 3(f) below, the Directors listed in column 1 of Table A:
 - shall be deemed to be either a Member Elected Director, Cricket Appointed Director or a Board Appointed Director on and from the Effective Date as indicated in column 2 of Table A against each of their names;
 - (ii) shall be deemed to retire or their appointment to office terminated as at the time listed in column 3 of Table A against each of their names (without prejudice to their opportunity to be re-elected or re-appointed to office, subject to the provisions of the Constitution).
 - (b) Any person holding office as a Director or President immediately prior to the Effective Date and whose name does not appear in Table A shall be deemed to have retired from such office as at the Effective Date.

Table A

		·
Name of Director	Member Elected Director (MD) or Cricket Appointed Director (CD) or	Deemed date of scheduled retirement from office
	Board Appointed Director (BD)	
Tom Percy QC	MD	AGM 2018
Michael Veletta	MD	AGM 2019
The Honorable Dr Ken Michael AC	MD	AGM 2019
The Honorable Terry Waldron	MD	AGM 2020
Darren Wates	MD	AGM 2020
David Bailey	BD	AGM 2018
Dr Vanessa Guthrie	BD	AGM 2019
Eva Skira	BD	AGM 2020
Avril Fahey	CD (SGDC)	AGM 2018
Christian Bauer	CD (WADCC)	AGM 2019

- (c) The persons holding office as Chair and Deputy Chair respectively immediately before the Effective Date will continue to hold such offices on and from the Effective Date until the first Board meeting following the next AGM held after the Effective Date, subject to the provisions of Article 6.3
- (d) The person holding office as Patron immediately before the Effective Date will hold office as Patron-in-Chief on and from the Effective Date, subject to the provisions of Article 6.1.

- (e) The persons holding office as Chief Executive Officer and Secretary, subject to paragraph 3(f) below, immediately before the Effective Date will continue to hold such offices on and from the Effective Date, subject to the respective provisions of Articles 6.4 and 6.5.
- (f) It is a condition precedent to each Director and the Secretary holding office as such on and from the Effective Date that they respectively have executed a consent or approval to act in such role in accordance with any regulatory requirements of the Act.
- (g) A Director listed in Table A holds office as such until the date mentioned in column 3 of Table A against their name notwithstanding Article 5.6.
- 4. With respect to Membership
 - (a) Subject to the person applying for Membership in the relevant Membership class in accordance with this Constitution, Members holding the class of membership noted in column 1 of Table B immediately prior to the Effective Date shall be deemed to hold the class of membership noted in column 2 of Table B on and from the Effective Date until 30 June next following the Effective Date without any additional Fee or Fee adjustment on account of that change of class of Membership.

Table B

Pre-Effective Date Class of Membership	Membership Class on and from Effective Date
Ordinary Member	Full Member (Metro Status)
Concession Member	Full Member (Concession Status)
Club Playing Member	Full Member (Player/Administrator Status)
Country Member	Full Member (Country Status)
Junior Member	Junior Member
Club Member	Full Member (Metro Status)

- (b) Honorary Life Members, Lifetime Members and Perpetual Members will continue to retain their status as such on and from the Effective Date with all rights and entitlements as if they were Honorary Members on terms determined by the Board as shall be consistent with their original appointment as Honorary Life Members, Lifetime Members or Perpetual Members (as applicable).
- 5. With respect to Affiliated Organisations
 - (a) On the Effective Date the organisations listed in Table C are each deemed to be an Affiliated Organisation.
 - (b) Nothing by reason of this Constitution varies or affects any agreement between the Organisation that was in force and effect immediately before the Effective Date.

Table C

Details of Affiliated Organisations as at the Effective Date
WA DISTRICT CRICKET COUNCIL INC.
WA FEMALE CRICKET COUNCIL INC
COMMUNITY JUNIOR CRICKET COUNCIL (INC)
INDOOR SPORTS W.A. INCORPORATED
WA COUNTRY CRICKET BOARD (INC)
CRICKET WEST INC.

Western Australian Cricket Association Limited, By-Laws

Updated August 2020

1. Introduction

- 1.1 These By-Laws have been adopted by the Board pursuant to Article 10.4 of the Constitution.
- 1.2 These By-Laws have effect from the Effective Date.
- 1.3 Should any By-Law conflict with a provision of the Constitution, then in the terms of Article 10.4 of the Constitution, that By-Law will be either void or read down to the extent of such conflict, and the provisions of the Constitution will prevail.
- 1.4 Words and phrases defined in the Constitution and used in these By-Laws have the meanings accorded to them in the Constitution
- 1.5 Where a power is conferred on the Board under these By-Laws, the Board may resolve to delegate that power to the CEO.
- 1.6 In these By-Laws the following words have the meanings set opposite them:
 - "Applicable Form" means a form referred to in these By-Laws which has been approved by the Board for the relevant functional purpose in these By-Laws.
 - "Concession Status" means a Full Member who qualifies for and has been conferred that status under By-Laws 3.3(f) and 3.6(a);
 - "Country Status" means a Full Member who qualifies for and has been conferred that status under By-Laws 3.3(d) and 3.6(a).
 - "Cricket Club" means an organisation (incorporated or unincorporated) which organises, co-ordinates and administers games of cricket for the benefit of its members.
 - "Facilities" means premises, grounds, rooms, areas and related facilities at grounds or areas under the control or licence of the Organisation.
 - "Metro Status" has the meaning ascribed to it in By-Law 3.3(q).
 - "Player/Administrator Status" means a Full Member who qualifies for and has been conferred that status under By-Laws 3.3(e) and 3.6(a).
 - "Guest Pass" means a pass referred to in By-Law 7.3(a).
 - "Tier Package" means an opportunity that may be offered by the Organisation in accordance with By-Law 8.
 - "Transferrable Associate Pass" means a transferable pass, which provides admission to designated "Member Reserved" areas at Facilities, but no other rights associated with Membership.
 - "Visitor Pass" means a pass referred to in By-Law 7.3(b).

2. Governance Charter

- 2.1 The terms of the Governance Charter adopted by the Board and prevailing as at the Effective Date apply on and from the Effective Date as having the force and effect of By-Laws duly adopted by the Board.
- 2.2 A copy of the Constitution, these By-Laws and the Governance Charter are to be available for inspection and reference by Members and others including on the Organisation's website.

3. Membership

- 3.1 (a) Membership of the Organisation is available in the following classes:
 - Full Member
 - (i) Metro Status
 - (ii) Country Status
 - (iii) Concession Status
 - (iv) Player/Administrator Status
 - Junior Member
 - · Honorary Member

Without prejudice to the continuing recognition of Honorary Life Members, Lifetime Members and Perpetual Members under Schedule 3 of the Constitution (Transitional Arrangements).

- (b) For the purposes of Article 2.1(b) of the Constitution, the Board by Absolute Special Resolution may determine any limits on the number of Members in a Class of Membership or limits on the number of Members within a status category of Full Membership.
- 3.2 The rights, privileges and obligations attaching to each Class of Membership are as follows:
 - (a) Full Member

Subject to payment of all applicable Fees in respect of the previous cricket season (or for new Members, by 31 August) (including any discounted Fees applicable to Full Members qualifying for Country Status, Concession Status or Player/Administrator Status) from time to time levied with respect to that Membership, a Full Member is entitled:

- (i) to vote, and to participate and speak (in accordance with rulings of the meeting chair) at meetings of Members and meeting of that Class of Member. For General Meetings other than the AGM, all applicable fees must have been paid by no later than one month before the meeting for a Member to be eligible to vote;
- to nominate a person as a new Member to the Organisation and to nominate a qualifying Member for consideration for election as a Member Elected Director;
- (iii) to stand for office as a Director:
- (iv) to admission, in common with other Members, to designated "Member Reserved" areas at Facilities subject nevertheless to:
 - (a) prudential crowd control limitations set by the Organisation or those in charge of the Facilities;
 - (b) payment of any applicable supplementary Fee (if any) determined from time to time by the Organisation as applicable to those Facilities:
 - (c) observance by the Member of all rules and codes of conduct prescribed by the Organisation or by those in charge or the Facilities, applicable to attendees at the Facilities;
 - (d) specifically no entitlement to admission arises with respect to games of cricket or events staged or held under the authority of the International Cricket Council ("ICC");

- (v) to receive all information required by the Constitution, these By-Laws or the Act to be forwarded to Members of that Class of Membership:
- (vi) to receive such publications and communications as the Organisation may from time to time determine is appropriate or desirable to forward to Members of that Class of Membership;
- (vii) to receive offers for Tier Packages that may be offered from time to time;
- (viii) to exercise such other rights and privileges determined from time to time by the Board to be accorded to Members of that Class of Membership.

(b) Junior Members

Subject to payment of all applicable Fees from time to time levied with respect to that Membership, and subject to the reservations and exclusions below, a Junior Member is entitled to all the same rights and privileges set out in By-Law 3.2(a)(i)-(viii) above for a Full Member OTHER THAN:

- voting rights under By-Law 3.2(a)(i), but without prejudice to any "Class" voting rights applicable for meetings of Members of that Class of Membership only;
- (ii) nomination rights under By-Law 3.2(a)(ii);
- (iii) unless the Junior Member shall have attained 18 years of age, access to certain Facilities which are available only to persons over 18 years of age under any applicable regulatory requirement including to do with liquor licensing and gaming;
- (iv) entitlement to stand for office as a Director under By-Law 3.2(a)(iii);
- (v) being obliged to allow Full and Honorary Members priority admission and seating at any "Member Reserved" areas at Facilities as referred to in By-Law 3.2(a) above.

(c) Honorary Members

Subject to any conditions of Membership as are determined from time to time by the Board to apply to the Honorary Member, an Honorary Member is entitled to all the same rights and privileges set out in By-Law 3.2(a)(i)-(viii) above for a Full Member, plus:

- Two complimentary transferrable guest passes to access the "Member Reserved" areas at Facilities;
- · Complimentary match-day parking.

(d) Perpetual Members

Subject to any conditions of Membership as are determined from time to time by the Board to apply to the Perpetual Member, a Perpetual Member is entitled to all the same rights and privileges set out in By-Law 3.2(a)(i)-(viii) above for a Full Member, plus:

- Two complimentary transferrable guest passes to access the "Member Reserved areas at Facilities;
- Complimentary match-day parking,
- Eligibility to transfer membership in perpetuity to a family member or close relative.
- 3.3 Membership of the respective Classes of Membership is available to the following persons who have completed all administrative requirements as set out in the By-Laws for their application and approval of Membership subject to the following:
 - (a) All Classes of Membership (other than Junior Membership) are only available to persons 18 years of age or over:

- (b) Junior Membership is only available to persons under 18 years of age as at the start of the relevant membership period;
- (c) Honorary Membership is only available to persons nominated by the Board for such Membership and who in the opinion of the Board have provided outstanding and special service to the Organisation or in the promotion, development and/or advancement of cricket:
- (d) Country Status for a Full Member is only available to persons whose principal place of residence is more than 100 kilometres from the regulatory designated place in Perth from which distances from Perth are customarily measured for official purposes;
- (e) Player/Administrator Status for a Full Member is available to persons who are current members of an Affiliated Organisation or a Cricket Club and who are:
 - Active in the management or administration (including voluntary) of that Affiliated Organisation or Cricket Club.
 - (ii) A Past Player who has represented WA at first class level or other elite level is eligible to apply for a single complimentary pass to access the "Member Reserved" areas at Facilities and benefits as advised from time to time. There are no voting rights associated with this pass.
- (f) Concession Status for a Full Member is only available to persons over 65 years of age;
- (g) Metro Status for a Full Member applies to a Full Member who has not been afforded Country Status, Player/Administrator Status or Concession Status;
- (h) Junior Members, who in due course qualify for admission as a Full Member, may transfer to becoming a Full Member by complying with By-Law 3.7.
- All WA Premier Cricket Umpires who are Members of WACUA and have umpired at least 8 matches in the previous season are eligible to the following benefits:
 - (i) Waiver of any waitlist or nomination fees;
 - (ii) Eligible to purchase a concession Membership;
 - (iii) After 5 consecutive years and having umpired at least 8 matches in the previous season, will be provided a complimentary Membership for as long as they are current umpires and have indicated their availability for the coming season.
- Current WACA staff and players are eligible to receive two complimentary Transferrable Associate Passes
- 3.4 All applications for Membership must be:
 - completed substantially in the Applicable Form relevant to the Class of Membership being applied for;
 - accompanied by such information as is required by the Organisation to reasonably verify that the applicant meets the requirements for Membership of the relevant Class set out in By-Law 3.3;
 - accompanied by payment of any Fees applicable to that application and Membership; and
 - (d) signed by the applicant;

PROVIDED THAT the Board may at its discretion:

- (a) waive strict compliance with any aspect or aspects of the foregoing;
- (b) require the provision of such further information from the applicant, which it considers it is appropriate so to do in the interests of the Organisation and the integrity of its Membership.

- 3.5 Without limiting By-Law 3.4, the Applicable Form to be signed by an applicant under By-Law 3.4 must include an undertaking by the applicant, conditional upon the applicant becoming a Member:
 - (a) in the terms of Article 1.2(c) of the Constitution; and
 - (b) to observe and comply with the provisions of the Constitution and the By-Laws to the extent to which they have application to a person who is a Member of the Class of Membership applied for.
- 3.6 (a) A Full Member who believes they qualify for Country Status, Concession Status or Player/Administrator Status may apply for recognition of that status by completing and signing the Applicable Form which may then be approved of by the CEO (acting under deemed delegate authority of the Board).
 - (b) A Full Member who has been conferred Country Status, Concession Status or Player/Administration Status but no longer qualifies for such status must promptly inform the Organisation in writing by completing and signing the Applicable Form which may then be approved of by the CEO (acting under deemed delegated authority of the Board).
- 3.7 A Junior Member (upon attaining 18 years of age) may apply for transfer of Membership to a Full Member by completing and signing the Applicable Form which may then be approved of by the CEO (acting under deemed delegated authority of the Board).

4. Fees and Interest

- 4.1 Fees may be for all or any of the following and in such amounts as determined from time to time by the Board, including any discount in Fees applicable to Full Members holding Concession Status, Country Status or Player/ Administrator Status in accordance with these By-Laws:
 - (a) initial Membership waitlist fee;
 - (b) initial Membership nomination fee;
 - (c) Membership annual subscription fee;
 - (d) Member Guest Pass fee;
 - (e) Member Visitor Pass fees;
 - (f) Other (as determined by the Board).
- 4.2 All Fees must be paid by the date notified in the Organisation's invoice for the same, or if no date is notified, within a reasonable time of the Organisation's invoice for the same.
- 4.3 As at and from the Effective Date, the Fees are the same as those applying immediately prior to the Effective Date.
- 4.4 Subject to By-Law 4.5, around the commencement of each Financial Year after the Effective Date, the Board by resolution duly passed, after consultation with management as part of the Organisation's budgetary process, will set the Fees then applicable for the forthcoming Financial Year and will communicate the same to the Members by notice under Article 11 of the Constitution.
- 4.5 Other than with the approval of the Members in general meeting, no Fee for annual Membership subscription may be increased under these By-Laws by more than 20% of the amount of the Fee that applied in the Financial Year immediately prior to the proposed increase.
- 4.6 Unless payment is waived under Article 3.2(b) of the Constitution, the prescribed rate of interest in the terms of Article 3.2 of the Constitution is 10% per annum simple interest.
- 4.7 With respect to replacing a lost Membership Card (refer Article 2.8 of the Constitution) the prescribed Fee is such amount reasonably determined by the Organisation.
- 4.8 Should a Member anticipate being unable to enjoy their Membership, through absence outside Western Australia

- or significant ill health, for a period exceeding one (1) year and not exceeding three (3) years, then that Member may make application to the Organisation in the Applicable Form for exemption from Fee payment and voluntary suspension of Membership rights during that period, whereupon the Organisation may at its discretion, and subject to any terms it may decide to impose, allow such interim Fee exemption and voluntary suspension.
- 4.9 No nomination fee is payable by a Junior Member applying to transfer to Full Membership under By-Law 3.7.
- 4.10 If any GST is chargeable in connection with a Fee, then unless stated to the contrary at the time of determination of that Fee, the amount of that GST is to be included in the Fee and is to be payable by the Member or person to whom the Organisation's tax invoice for the Fee is rendered.
- 4.11 Should a Member's annual fees not be received by the Organisation for a period exceeding 24 months from the scheduled due date under By-Law 4.2, all benefits associated with continuous longevity of Membership will cease until all applicable fees, including nomination fees, are paid for the period lapsed.

5. Applicable Forms

- 5.1 The management of the Organisation may from time to time propose to the Board Applicable Forms in order to best manage and administer functions to be undertaken in the terms of these By-Laws.
- 5.2 The Board may by resolution approve, with or without amendment, any such proposed Applicable Forms.
- 5.3 Upon such approval by the Board, the relevant form will then be the prescribed Applicable Form for the purposes of these By-Laws and the Constitution (as applicable).
- 5.4 Without limitation, Applicable Forms may relate to any or all of the following functions and requirements:
 - (a) Constitution
 - (i) Article 2.2 Application
 - (ii) Article 2.6 Expulsion or Suspension
 - (iii) Article 2.8(b) Declaration of Lost Membership Card
 - (iv) Article 4.14(f) & (j) Proxy form
 - (v) Article 5.3(b) & (c) Nomination of Member Elected Director
 - (vi) Article 5.11 Director's consent and undertakings
 - b) By-Laws
 - (i) By-Law 3.4 Application for Membership
 - (ii) By-Law 3.6(a) Country Status, Concession or Player/Administrator Status
 - (iii) By-Law 3.6(b) Cessation of designated status
 - (iv) By-Law 3.7 Transfer from Junior Membership
 - (v) By-Law 4.8 Temporary Absentee Member Fee waiver and voluntary suspension

5. Miscellaneous

- General conduct of meetings of Members (Article 4.8)
 No special By-Laws apply.
- 6.2 Nomination of Member Elected Directors (Article 5.3(c)(i))
 - (a) Refer By-Law 5.4(a)(v) and (vi) for the Applicable Form
 - (b) Refer By-Law 2 Governance Charter
- 6.3 Presentation of nominations for Member Elected Directors to Membership (Article 5.3(c)(ii))
 - (a) The Board will approve at its discretion the manner in which it is proposed for nominations for Member Elected Directors to be presented to Members.

- The Board will act impartially and in good faith in exercising its discretion with a view to the Members being presented with adequate, timely but succinct information to enable them to make a reasonably informed judgement when exercising their vote.
- (b) Nominees for Member Elected Directors may provide a brief statement to the Organisation concerning their candidature for office at least 21 days before the scheduled date of the issue of the notice of meeting and election/voting papers to Members provided the Board is satisfied that the statement is not offensive to, or likely to offend, the Organisation or any Member of it. If the Board is not so satisfied the Board reserves the right to censor any portion of the statement that the Board (in good faith) considers to be offensive to, or likely to offend, the Organisation or any Member of it.
- 6.4 Manner in which voting for Member Elected Directors is to occur (Article 5.3(c)(iii))
 - (a) Subject to the Act, the Board will approve at its discretion the manner in which voting for Member Elected Directors is to occur.
 - (b) The Board will act impartially and in good faith in exercising its discretion to ensure integrity and probity of process.
 - (c) In exercising its discretion, the Board may place due reliance on advice and recommendations from its legal representatives and/or its Member registry provider as to what is good governance practice for such processes for a membership organisation with a membership base of the size of the Organisation.
- 6.5 Directors Consents and Undertakings (Article 5.11) Refer By-Law 5.4(a)(vi) for the Applicable Form.
- 6.6 Role of Patron-in-Chief (Article 6.1(d))
 - (a) The role of the Patron-in-Chief will largely be ceremonial under the guidance of the Chair.
 - (b) Subject to the Patron-in-Chief's availability and to other commitments that the Patron-in-Chief may have, the Patron-in-Chief may be invited by the Chair and/or the CEO on behalf of the Organisation to attend, and as appropriate to officiate and speak as Patron-in-Chief, at important cricket matches, general meetings of Members and significant occasions and events promoted by the Organisation.
- 6.7 Role of Patron(s) (Article 6.2(e))
 - (a) The role of the Patron(s) will largely be ceremonial and promotional under the invitation and guidance of the Chair and/or the CEO.
 - (b) Subject to their availability and other commitments they may have, it is expected that between them, the Patron(s) will be available at the invitation of the Chair and/or the CEO on behalf of the Organisation to co-host and perhaps speak at promotional events organised by the Organisation incidental to the game of cricket and cricket matches, and to attend meetings of Members, as ambassadors of the game of cricket.
- 6.8 Secretary (Article 6.5(c))
 - Refer to By-Law 2 Governance Charter.
- 6.9 Board Committees Terms of Reference and Governance Refer to By-Law 2 Governance Charter.

7. Membership Cards, Facilities Entry and Behaviour

7.1 Members may be issued by the Organisation a Member Card to identify that Member as a Member of the Organisation. Members must display their Member Cards at all times when exercising their Membership right to access Facilities.

- 7.2 No assurance is given by the Organisation with respect to the Facilities as to:
 - (a) the availability or suitability of seating;
 - (b) the suitability of viewing;
 - (c) the suitability of premises;
 - (d) the availability or suitability of services and amenities.
- 7.3 (a) A Member, on payment of an annual Guest Pass Fee(s) as determined from time to time by the Board, is entitled to receive Guest Passes which that Member may pass to any person, and which entitles the holder thereof for the time being whilst wearing or displaying the Guest Pass, the right to be admitted to those Facilities (or designated parts thereof) allocated for the use of Members. The Organisation is under no obligation to check whether the holder of a Guest Pass is the person whom the Member intended to be the user thereof. Guest Passes are personal to the Member to whom it has been issued and such a Member must not transfer, assign or on-sell Guest Passes to another person(s). A guest to whom a Guest Pass has been passed by a Member may not sub-delegate the use of that Guest Pass to another person, other than with the express authority of the Member as to the identity of that other person. The Organisation may allow a Member to acquire more than one Guest Pass.
 - A Member, on payment of Visitor Pass Fees applicable to the designated duration, games or events the subject of a relevant Visitor Pass offered from time to time by the Organisation may receive Visitor Passes which that Member may pass to any person, and which entitles the holder thereof for the time being whilst wearing or displaying the pass, the right to be admitted to those Facilities (or designated parts thereof) allocated for the use of Members which are within the scope of the terms of issue and use of the Visitor Pass. The Organisation is under no obligation to check whether the holder of a Visitor Pass is the person the Member intended to be the user thereof. A Member must not transfer, assign or on-sell a Visitor Pass. The Organisation may allow a Member to acquire more than one Visitor Pass.
 - (c) A Member to whom Guest Passes or Visitor Passes are issued is responsible for ensuring that the holders of the Guest Passes or Visitor Passes (as applicable) observes the requirements of By-Law 7.8 while the holder is on any part of the Facilities.
 - (d) A Transferrable Associate Pass includes the same restrictions as described under By-Laws 7.3 (a) to (c).
- 7.4 A Member shall not transfer, intentionally or otherwise, a Member Card or allow any person, including family members or friends or associates to use the Member Card. The Member is responsible to ensure that their Member Card is not used by any other person.
- 7.5 A Member shall notify the Organisation of the loss or theft of a Member's Member Card, Guest Card and/or Visitor Pass as soon as reasonably practicable after becoming aware of its loss or theft.
- 7.6 A Member Card, Guest or Visitor Pass which is used by any person not entitled to use it to gain or attempt to gain access to any Facilities, may be retained on behalf of the Organisation by a person duly authorised by the Organisation who comes into possession of that Member Card.
- 7.7 Upon demand by a duly authorised person on behalf of the Organisation:
 - a Member seeking entry to Facilities shall produce their Member Card and supporting photographic identification;

- (b) a holder of a Guest Pass or Visitor Pass seeking entry to Facilities shall produce their Guest Pass or Visitor Pass and inform the duly authorised person the name of the Member who gave them the Guest Pass or Visitor Pass. The relevant Member must verify (or otherwise) that the person is their guest or visitor on request by the duly authorised person;
- (c) for the purposes of By-Law 7.7(a), "supporting photographic identification" means a valid driver's licence, passport or other identification to the reasonable satisfaction of the duly authorised person.
- 7.8 Each Member, Guest Pass holder, Visitor Pass holder, Transferable Associate Pass holder and any other visitor of the Member using Facilities must at all times:
 - (a) comply with any Lawful direction or request of the Organisation or any agent, employee, contractor or authorised officer of the Organisation or member of the police force while at, or using, the Facilities including a direction or request to vacate and leave the Facilities;
 - (b) not engage in conduct that:
 - is unbecoming or prejudicial to the interests of the Organisation; or
 - (ii) may bring the Organisation into disrepute; or
 - (iii) may be such that impairs or is likely to adversely or affect the reasonable enjoyment of the benefits of Membership by other Members or their quests or visitors; and
 - (c) be subject to the applicable dress requirements set down for the relevant Facilities as determined from time to time and advised in Member communications or signage in or around the Facilities.
- 7.9 The Member is responsible for ensuring that their Guest Pass holder and Visitor Pass holders comply with By-Laws 7.3(a), 7.7 and 7.8. The Member shall be held accountable for any breach of these By-Laws by their Guest Pass holder or Visitor Pass holder.

8. Tier Packages

- 8.1 The Organisation may offer to Members upon terms and conditions determined from time to time by the Organisation, including as to Fees payable, the opportunity to have access to and/or seating in designated reserved areas of Facilities which are not otherwise generally available to Members as an ordinary entitlement of Membership.
- 8.2 Members to whom an offer for a Tier Package is made may accept the offer by paying the applicable Fee and satisfying any other terms of the offer, which will constitute an agreement by the Member to observe the terms and conditions upon which the relevant Tier Package is offered.
- 8.3 A Tier Package may include rights for the Member to be accompanied in the designated reserved area of Facilities by one or more guests.
- 8.4 The Organisation may issue to any Members taking up a Tier Package, and to each of their authorised guests (if provided for in the terms of the Tier Package), Tier Package passes ("TP Passes") which must be worn by the Member and such guests (if applicable) at all times while accessing and exercising their rights and privileges in the terms of the relevant Tier Package.
- 8.5 Any Member or other person purporting to be entitled to exercise rights or privileges under a Tier Package must:
 - produce to a duly authorised person on behalf of the Organisation their TP Pass for verification;
 - (b) comply with By-Law 7.8.

8.6 The Member shall be held accountable for any breach of this By-Law 8 by any guest of the Member to whom a TP Pass may have issued or been given.

9. Members' access to the register of Members

- 9.1 A register of Members ('register') must be maintained and must include the name and address of all Members and the date on which they became a Member.
- 9.2 Although the Privacy Act 1988 (Cth) applies to personal information in the register, that information may be disclosed by the Organisation pursuant to the Privacy Act if required by law (including the Corporations Act).
- 9.3 The Organisation is a public company limited by guarantee and must comply with the Corporations Act 2001 (Cth) ('Corporations Act').
- 9.4 Pursuant to the Corporations Act, anyone, including non-Members, is allowed to inspect the register without the need to provide reasons.
- 9.5 A person may obtain a copy of the Organisation's register provided that the person wishing to obtain a copy (the "applicant") must first apply in writing to the WACA CEO to do so.
- 9.6 Member email addresses and phone numbers will not be included in the register to be inspected or copied.
- 9.7 The applicant will be allowed to obtain a copy of the register if:
 - the application states the purposes for which the person wishes to obtain the copies;
 - · None of the reasons are "prescribed purposes";
 - · The applicant pays any fees required; and
 - The application includes the person's name and address.
- 9.8 After obtaining a request, a decision on how to respond will be made by the Organisation within seven days.
- 9.9 Members and non-Members may be charged a reasonable fee for being provided with a copy of the register.
- 9.10 Members may inspect the register without being charged a fee, however non-Members may be charged a reasonable fee to inspect the register.
- 9.11 An application to inspect or obtain a copy of the register may be refused if the fee is not paid.

This page is blank intentionally.











Western Australian Cricket Association Limited

ABN 44 026 744 769 | ACN 627 195 402 WACA Ground, Nelson Crescent, East Perth, WA 6004 PO Box 6045, East Perth, WA 6892

P: +61 8 9265 7222 E: info@waca.com.au